Machine Translated by Google

BULKPDF BUSINESS

DOCUMENTATION



VERSION 2024.09.24.1311

BULKPDF.COM

COPYRIGHT ©2024 TADELSUCHT UG (LIMITED LIABILITY)

1 Introduction	5
1.1 What is BulkPDF and how does it work? .	5
1.2 Functions .	6
1.3 Requirements	9
1.3.1 System	9
1.3.2 Data source - spreadsheet file .	9
1.3.3 Data source - Microsoft SQL Server	
1.3.4 Data source – Microsoft Access .	
1.3.5 JSON .	11
1.3.6 XML	11
1.3.7 Data source - MySQL / MariaDB Server	12
1.3.8 PDF form .	12
1.4 Standard PDF form field types and corresponding data	
1.4.1 Text box	
1.4.2 Check box	
1.4.3 Radio button .	14
1.4.4 List box	14
1.4.5 Combo box	14
1.5 BulkPDF Business special field types and corresponding data 15	
1.5.1 Hyperlink	
1.5.2 Image	
1.5.3 File attachment	
1.5.4 Additional buttons for filling in scanned	
Forms	
1.5.5 Additional "text box" .	
1.5.6 Additional "radio button"	
1.5.7 Additional "Checkbox button" .	
1.5.8 Barcodes	
1.6 Installation & Licensing	
1.6.1 Graphical installation .	
1.7 Unattended installation (Silent installation) .	
1.7.1 Licensing	
1.7.2 Multi-user / Terminal Server Licenses	
2 Application scenarios (step by step explanation)	20
2.1 Supplied sample files	
2.1.1 Example files in detail	
2.1.2 Filling out the sample PDF form	22
3 User interface in detail (GUI)	32

3.1 Home	32	
3.1.1 Overview	33	
3.1.2 License activation	34	
3.1.3 Configuration: Save & Load .	35	
3.2 Data sources .	36	
3.2.1 Overview	36	
3.2.2 Excel / spreadsheet files	37	
3.2.3 Microsoft SQL Server .	39	
3.2.4 Microsoft SQL Server Custom Query .		41
3.2.5 Microsoft Access		42
3.2.6 Microsoft Access Custom Query		44
3.2.7 JSON .	45	
3.2.8 XML	47	
3.2.9 Spreadsheet / Excel (Swapped columns and rows) 48		
3.2.10 MySQL/MariaDB Server .	48	
3.2.11 MySQL/MariaDB Server Custom Query .	50	
3.3 PDF form	51	
3.3.1 Overview	52	
3.3.2 PDF preview .	54	
3.3.3 Assigning data to PDF form fields .	55	
3.3.4 Textbox & General Field Data Selection .	56	
3.3.5 Using a data source for a form field	58	
3.3.6 Set constant text for form field .	60	
3.3.7 Write protection on individual form fields	61	
3.3.8 Date formatting in form fields	62	
3.3.9 Check box	64	
3.3.10 Radio buttons	65	
3.3.11 List box		
	67	
3.3.13 Barcodes in general	07	
3.3.14 QR code	00	
3.3.15 Hyperlink	70	
3.3.16 Figure	/ 1	
3.3.17 File attachment	72	
3.5 Completion	۰۰۰۰، ۱۱ مم	
3.6 Email Draft Generator	۵۵ ۵0	
3.7 Miscellaneous	04 0 <i>F</i>	
3.7.1 Support window	oo	

3.7.2 Documentation window .				8	36
3.7.3 Error window (Exception).				8	37
3.7.4 Font selection & font color	selection			8	37
3.7.5 Additional functionalities				8	38
4 Command line (BulkPDF Business Co	onsole)				90
4.1 Parameter list				9	90
4.2 Output from the console application	n			(92
5 Configuration files					96
5.1 Relative paths				9	96
5.2 Example excerpts from a configura	ation file .			9	97 28
					20
5.2.2 Special Fields .					20
5.2.3 Global options 5.2.4 FilenameGenerator					100
5.2.5 Email					102
0.2.0 Email					102
6 tips and tricks					103
6.1 Create or repair PDF forms with PI	OFescape .				103
6.2 Creating PDF forms with LibreOffic	e / OpenOf	fice Writer	103		
6.2.1 Text field example (text bo	x)			· · · · · · · · ′	104
6.2.2 Radio button example				· · · · · · · · ′	105
6.2.3 Checkbox .				· · · · · · · · ′	106
6.2.4 List box				′	107
6.2.5 Combo box .				′	108
6.3 Create signature				′	108
6.4 Date formatting .				· · · · · · · · ′	108
6.5 Connection string					111
6.6 Simultaneous printing of document	IS .				111
7 Frequently Asked Questions (FAQ)					112
7.1 What are the system requirement	its for Bulkl	PDF Busin	ess?		112
7.2 BulkPDF cannot be installed. [Wind	dows Smart	Screen]			112
7.3 If my files or data are uploaded or	left in				
my computer in any way? .				′	113
7.4 Is the program available for Mac O	S?			· · · · · · · · · · ·	113
7.5 How many users may use a license	e simultane	ously?			
the?				· · · · · · · · · · ·	113
7.6 On how many computers can a lice	ense be use	ed? 11	3		

7.7 I get the following message "The license key is currently already used by several devices". What does that mean? 113
7.8 In the PDF preview, the form fields are not displayed in the correct
gen position or shifted
7.9 No PDF form fields are displayed in BulkPDF Business114
7.10 Unicode characters in Acrobat Reader are not displayed correctly
(Chinese, Cyrillic, Arabic, etc)
7.11 Windows always opens the wrong PDF viewer or
Web browser (Chrome, Edge) for viewing PDF files 115
7.12 System.InvalidCastException: Could not convert data to boolean 116
7.13 System.IndexOutOfRangeException: "Only X radio buttons in the group
found. Selected index was Y."
7.14 System.InvalidOperationException: The 'Microsoft.ACE.OLEDB.12.0'
provider is not registered on the local machine
7.16 The date field is additionally specified with 00:00:00. How can that be removed?
7 17 L baye my data source files (Excel CSV etc.) or my
PDF file moved. Are my BulkPDF business
configuration file?
7.18 When calling the console application via a batch script,
the characters, such as German umlauts, are not correctly transmitted 117
7.19 Combinations of multiple columns for one field
7.20 "Checkbox" or "Radio button" should be according to data and preview in
the XFA PDF, but are not in Acrobat Reader 117
7.21 How can I correct the number format in my generated PDF?
It does not maintain the format set in the table. How can
For example, numbers with two decimal places and a EUR symbol
to display at the end? 118

8 References

Introduction

1.1 What is BulkPDF and how does it work?

In short: BulkPDF automatically detects the available data in

the selected tabular data source (Excel, SQL databases, etc.). The

found column names can then be assigned manually and/or automatically to the form fields. After filling has started,

The software scans the table line by line and inserts the cell value into the PDF form. The PDF file is then opened with the cell value also taken from the table.

saved under the selected file name. You will then receive a completed PDF form that is ready to be sent.



1.2 Functions

Below is a brief overview and a detailed list of the program's functions.

Brief overview

Easy to use No expert knowledge required, clear and easy to use.

Optimize for the future Once created, a PDF configuration can be quickly be reused.

Data source support Various data sources are supported (For example: Excel, CSV, SQL databases, ...)

Data protection Without cloud. Without uploading your data. In compliance with European data protection regulations. Made in Germany.

Updates The program is constantly being developed and new features equipped.

Automation in just a few steps

• 1. Select data source

- 2. Assign data to the PDF form fields, manually and/or automatically table
- 3. Select file name

Create hundreds of PDF documents at once Create any number of completed PDF documents at once with one click.

Barcodes Various barcodes are supported. (For example: QR, Data-Matrix, PDF417, EAN8, EAN13, Codabar, Code11, Code39, Code93, Code128[A,B,C], GS1-128, UPC)

Additional features

- AcroForm and XFA support
- PDF signatures
- Data dependent file names
- Write protection
- Images and hyperlinks
- PDF file

Detailed feature list

- Software License: Proprietary
- Automatically fill out PDF forms
- No watermark (outside DEMO)
- Unlimited number of PDF files
- Modern, clear interface
- Active development
- Live PDF preview
- Sign PDF files
- Microsoft Excel
- LibreOffice / OpenOffice

- Microsoft SQL Server
- MySQL / MariaDB
- Dynamic PDF file naming
- PDF compression
- Date formatting
- Pictures
- Hyperlink
- Email draft generator
- Unidimensional barcode (Codabar, Code11, Code128A, Code128, Code128B, Code128C, Code39, Code93, CodeUpc, Ean13, Ean8, GS1Code128)
- Bidimensional barcode (Pdf417, DataMatrix_Barcode, QR)
- Documentation: Yes, but still under development.
- Can be used without internet (offline): Yes, after a one-time online activation.
- Supports BulkPDF Freeware configuration files
- Supports BulkPDF Business configuration files
- Console application (batch script integration)
- Support for different writing systems (Unicode)
- Selection of good-looking fonts for different languages or Writing systems: In development.
- Custom fonts
- AcroForm
- XFA: Yes, but some XFA documents may not be supported.
- Windows
- Mac OS: No, but can be used with a virtual machine (For example: Parallels Desktop, VirtualBox)
- Linux: No, but can be used with a virtual machine (For example: VirtualBox)

1.3 Requirements

1.3.1 Systems

Compatible with Windows 7, Windows 8, Windows 10 and Windows 11. Approximately 500 MB of storage space is required for installation. The program requires a onetime Internet connection for license activation.

1.3.2 Data source - spreadsheet file

In order to use the data in the spreadsheet file, a certain The first line is always the column names. These should have a clear and understandable name because they are used in the program to select the desired data sets BulkPDF checks the column names until it finds an empty cell. Therefore, for example, the first row in the first column must not be empty, otherwise no further data can be found. The program then determines the maximum Number of rows in the table. The program goes through line by line and only stops when a row consists entirely of empty cells. For example, a valid table could look like this:

First name	Last Name Address		
Miller	Bernd	Far-Far-Away	
Mr. Smith		Mustermann Street	
Model Woman N	<i>l</i> arie	Mustermann Street	

For another example, the included sample spreadsheet file be considered (see 2.1.1).

Furthermore, there is a separate data source for reading in a spreadsheet file, where the behavior of columns and rows is swapped so that, for example, the following table can also be used (see section

3.2.9):

First name Be	ernd	Max	Mary	
Last name Mu	ueller	Mustermann	Model woman	
Address	Far-Far-Away Must	ermann Street Musterman	n Street	

It is worth noting that the use of Excel functions in the Excel data source and provides a useful way to store data

to process and prepare before inserting them into the PDF form field become.

File Type Container Format File Format .xlsx OpenXml Excel version(s)						
2007 and new	2007 and newerIP, CFB+ZIP					
.xlsb ZIP, CFB OpenXml 2007 and newer						
xls	CFB	BIFF8	97, 2000, XP, 2003			
	0.0	Birro	98, 2001, vX, 2004 (Mac)			
.xls	CFB	BIFF5	5.0, 95			
.xls	-	BIFF4	4.0			
.xls	-	BIFF3	3.0			
.xls	-	BIFF2	2.0, 2.2			
.CSV	-	CSV	(All)			

The following spreadsheet file formats are supported by the program[7]:

1.3.3 Data source - Microsoft SQL Server

There are several options available for obtaining data from a Microsoft SQL Server.

On the one hand, you can choose the option where you simply enter a "connection string" and then automatically find all available

Tables are displayed. One of these tables can then be selected to act as a data source. For valid "connection string" examples, see

Please refer to section 6.5 .

The second option is the possibility, in addition to the "connection string", simply to enter an SQL query that will return the desired data. This Option allows the full functionality of T-SQL on the Microsoft SQL

Server for the data source within BulkPDF.

1.3.4 Data source - Microsoft Access

There are several options available for using data from a Microsoft Access server. Variants for data acquisition are available.

On the one hand, you can choose the option where you simply select a database file (.accdb or .mdb) and then all available tables for the respective database are automatically displayed. From these tables, you can then one must be selected to act as a data source. The second option is the ability to enter an SQL query next to the file, which gives you the desired data. This option allows the full

Functionality of SQL on the Microsoft Access database for the data source within BulkPDF.

In some cases it may be necessary to download and install missing dependencies, such as the "Microsoft Access Database Engine 2010 Redistributable 64 Bit Version", see point 7.14.

1.3.5 **JSON**

In order to use the data in a JSON file, the data must be in a specific format. The following JSON structure would be in the complete state

e.g. readable in BulkPDF:



1.3.6 XML

In order to use the data in an XML file, the data must be in a specific format. The following XML structure would be in the complete state e.g. readable in BulkPDF:



1.3.7 Data source - MySQL / MariaDB Server

Integrating data from a MySQL or MariaDB server into BulkPDF can be realized in several ways.

One of the possibilities involves entering a "connection string" through which automatically lists the available tables of the associated database. A table can then be selected from this list as a data source. For examples of valid "connection string" formats, refer to

Please refer to section 6.5.

Another way is to formulate a SQL query that provides the required data in addition to the connection string. This method allows the full range of SQL functions to be used on the MySQL or MariaDB.

Server can be used to configure the data source within BulkPDF. eren.

1.3.8 PDF form

A PDF form is a PDF file that contains form elements in the format of Acro-Forms or XFA. Such form elements can be recognized, for example, in Acrobat Reader, for example, if you click in a text field and enter a text If, on the other hand, in Acrobat Reader you have to first drag the frame of the text field itself in order to enter text, then It is not a PDF form in the true sense, but a PDF File that represents a form. This representation is then only a scanned image or a form marked with visible characters (e.g. underlines) that can only be filled out manually in the printed version. Such documents are only supported to a very limited extent by BulkPDF Business with the additional "text box", "radio button" and "checkbox" special field types.

In a real PDF form, all of these form fields have a name and possibly a value or a default value. To make BulkPDF easier to use, the form field names in the PDF file should be descriptive so that

these can be easily identified when assigning the data in BulkPDF.

For a valid example, see the sample form provided with the program or Section 2.

1.4 Standard PDF form field types and matching data

This subsection describes all field types that can appear in a PDF form. It also describes the data that BulkPDF Business expects for the correct completion of the

Fields.

1.4.1 Text box

A "text box" field contains a text that can be single-line or multi-line, depending on the design. Furthermore, the font, font size and write protection can be set for such a field. Based on the selected font,

all conceivable characters are supported.

TEXTFELD-BEISPIEL (TEXTBOX):

Vorname:

Nachname:

1.4.2 Check box

A checkbox field can be selected or deselected.

MARKIERFELD (CHECKBOX):

Umweltfreundliche Verpackung:

The following values can be used to fill in this field type:

"X" Selected "	" Not selected
х	
1	0
true	false
on	off
у	n
yes	no
Yes	no
yes	not
sí	

1.4.3 Radio button

A radio button is actually a group of selectable checkbox-like fields, but only one of these elements can be selected.

To fill in this field type, you must specify a number that corresponds to the index of the element. In the following example, "Male" would be the value "1", "Female" the value "2" and "Diverse" the value "3".

OPTIONSFELD-BEISPIEL (RADIOBOX):

Geschlecht:

Männlich 🔿 Weiblich 🔿 Divers 🔿

1.4.4 List box

A "Listbox" is a list field type where one or more values can be selected from the list, depending on the PDF form settings.

The values specified by the creator of the PDF form can be filled in

whether via a constant value or dynamically via a data source. In the dynamic variant, the text to be selected must be as in the

PDF form must be written out.

LISTENFELD (LISTBOX):

Zugehörige Branche:	IT Industrial Cake-Factory Tax
---------------------	---

1.4.5 Combo box

A "combo box" behaves largely like a "list box" with the difference that this field type is intended to allow you to enter your own non-predefined

values, can use.

KOMBINATIONSFELD (COMBOBOX):

Derzeitige Tätigkeit:

1.5 BulkPDF Business special field types and matching data

1.5.1 Hyperlink

A hyperlink consists of two components, the text and the URL to which it is to be linked. Both values must be specified next to the position of the field. You can also set a font for a hyperlink.

1.5.2 Image

In addition to a position and the desired size, an image requires a file path as a value. The following file types are supported:

- 1. BMP
- 2.JPEG
- 3. GIF
- 4. PNG
- 5. TIFF
- 6. ICO & ICON

1.5.3 File attachment

A file attachment has a path to a file on the system as a value. The description of the file is the file name and the MIME type is automatically determined based on the file extension.

1.5.4 Additional buttons for filling out scanned forms

If it is not a PDF form in the true sense, but to create a PDF that only contains images of forms, there is no Ability to edit the form fields as you would with a real PDF form field. In BulkPDF Business, however, you can add additional fields

over it and of course fill it with data. The form fields shown in the PDF file can be visually overlaid by using the following

The more precisely explained "Textbox", "Radio button" and "Checkbox button" are placed appropriately above it.

1.5.5 Additional "text box"

An additional "text box" corresponds to a "text box" that is read-only and can be added to the PDF document using BulkPDF without the presence of an actual "text box" field in the PDF document. In addition to the normal necessary settings, the position and size of the field are also required.

1.5.6 Additional radio button

An additional "radio button" corresponds to a "radio button" that is read-only and can be added to the PDF document using BulkPDF without the presence of an actual "text box" field. In addition to the normal necessary settings, the position and size of the field are also required.

The field is a single radio button, so if you want to have a group of elements, you need to add several of these special field types.

1.5.7 Additional "Checkbox button"

An additional "Checkbox button" corresponds to a "Checkbox button" that is read-only and can be added to the PDF document using BulkPDF without the actual "Checkbox button" field being present in the PDF document. In addition to the normal necessary settings, the position and size of the field are also required.

1.5.8 Barcodes

Barcode fields can be generated dynamically based on the data for each individual PDF form and are inserted as an image at the corresponding selected position in the form. The following is a list of the two barcode groups with all supported barcode types:

Bidimensional Barcode

- 1. PDF417: https://en.wikipedia.org/wiki/PDF417
- 2. DataMatrix: https://en.wikipedia.org/wiki/Data_Matrix
- 3. QR: https://en.wikipedia.org/wiki/QR_code

Unidimensional Barcode

- 1. Codabar: https://en.wikipedia.org/wiki/Codabar
- 2. Code11: https://en.wikipedia.org/wiki/Code_11
- 3. Code128, Code128A, Code128B, Code128C: https://en.wikipedia.org/wiki/ Code_128
- 4. Code39: https://en.wikipedia.org/wiki/Code_39
- 5. Code93: https://en.wikipedia.org/wiki/Code_93
- 6. CodeUpc: https://en.wikipedia.org/wiki/Universal_Product_Code
- 7. Ean8, Ean13: https://en.wikipedia.org/wiki/International_Article_ Number
- 8. GS1Code128: https://en.wikipedia.org/wiki/GS1-128

1.6 Installation & Licensing

Below you will find information regarding installation types and licensing types.

1.6.1 Graphical installation

The necessary steps for installation as screenshots:



1.7 Unattended installation (Silent installation)

The parameter "/S" allows an unattended installation. If the installation path should be specified, this can be done using the parameter $/D=C:\$ NewInstallDir $\$. For this type of installation,

For licensing, copy a "LicenseConfirmation" into the %appdata%\BulkPDF folder or copied into the installation path on the system. This "LicenseConfirmation" can be activated on any system via the activation in the program or via the "BulkPDF Business Terminal Licensor" as long as it is This is a multi-user license. Depending on the activation type, this is in the %appdata%\BulkPDF folder or in the installation path on the system. This is not possible for single-user licenses, as the "LicenseConfirmation" contains a fixed hardware ID and is checked at startup whether this still matches the executing system.

1.7.1 Licensing

Normal licensing is done via the program on the home page. (See chapter 3.1.2)

1.7.2 Multi-user / Terminal Server Licenses

To license a terminal server, the "BulkPDF-Business-Terminal-Licensor.exe" located in the installation folder must be called with elevated rights. After entering and activating a valid multi-user/terminal server license key, this application creates a license file in the installation folder itself. This license file is later used by BulkPDF Business for license verification for the terminal server users.



Application scenarios (Step by Step explanation)

The following application scenarios describe an example use of the Software.

2.1 Supplied sample files

When starting the program, the example files can be loaded via the information text on the first page of the program. The example files can also be loaded directly (see section 2.1.1).

2.1.1 Example files in detail

The example files consist of the configuration "ProcessExample.BulkPDF", from the data source "ProcessExample.xlsx" and the PDF form "ProcessEx-ample.pdf", which can be found in the installation folder ("C:\Program Files\BulkPDF Business", provided the installation path was not changed during installation).

BulkPDF Business Configuration: "ProcessExample.BulkPDF" This BulkPDF Business configuration file contains all the information that was used in a previous session as well as references to the individual files, such as data source ("ProcessExample.xlsx") and PDF form ("ProcessExample.pdf"). These The configuration can be loaded again and again to avoid having to make all the settings again and again. This file is encoded in JSON format[5], is reasonably readable for humans and can be edited manually with any text editor if necessary, which can be particularly necessary when manually adjusting file paths.

Data source (Excel file): "ProcessExample.xlsx"

The spreadsheet file consists of tables. The table "Table1 (Customer-

data)" which is selected in the example configuration and the table "Table 2 (creditworthiness)".

B22	~	$f_{\mathbf{X}} \Sigma =$	=							
	A	В	с	D	E	F	G	н	1	1
1	Nachname (Textbox)	Vorname	Straße	Hausnummer	Postleitzahl	Ort	Geschlecht (Radio Button / Radiobox)	Umweltfreundliche Verpackung (Checkbox)	Zugehörige Branche (Listbox)	Derzeitige Tatigkeit (Combobox)
2	Mueller	Bernd	Weit-Weit-Weg	23	101781	Berlin	1		1 IT	Angesteller
3	Mustermann	Max	Mustermann Straße	5	80331	München	1		0 Industrial	Ehrenamtlicher
4	Musterfrau	Marie	Mustermann Straße	7	315151	Nunstorf	2	Yes	Cake-Factory	Freelancer
5	Klein	Berthold	Kleiner Weg	3	70173	Stuttgart	3	No	Tax	König von Deutschland
6	Groß	Anna-Lisa	Blumenweg	13	402101	Düsseldorf	3	Ja	Top secret	Ehrenamtlicher
7	蛋糕	蛋糕	饼干	17	34117	Kassel	2	是	Industrial	Ehrenamtlicher
8	торт	торт	печенье	2	391041	Magdeburg	1	нет	Cake-Factory	Angesteller
9	ł, ł, é, č	הקליטה	いろは	37	24103	ป็นหนุ	1	sí	IT	Angesteller
10										
11										
12										
10										

D9	<u>∽</u> Jx ∠ - =							
	A	В	с					
1	Nachname (Textbox)	Vorname	Kreditwürdig					
2	Mueller	Bernd	Yes					
3	Mustermann	Max	y					
4	Musterfrau	Marie	x					
5	Klein	Berthold	oui					
6	Groß	Anna-Lisa	sí					
7	蛋糕	蛋糕	是					
8	торт	торт	да					
9	ř, ľ, ě, č	הקליטה	true					
10								

PDF form: "ProcessExample.pdf"

The sample PDF form contains all standard PDF form elements so that you can fill them out once when testing the program.

Beispiel-Bestell-Formular
[Company Name] [Street Address, City, ST ZIP Code] [Phone] [Email] [Website]
TEXTFELD-BEISPIEL (TEXTBOX): Vorname: Nachname:
Straße: Hausnummer:
Postleitzahl: Ort:
OPTIONSFELD-BEISPIEL (RADIOBOX):
MARKIERFELD (CHECKBOX):
Umweltfreundliche Verpackung:
Zugehörige Branche: Tax
Derzeitige Tätigkeit:

2.1.2 Filling out the sample PDF form

Following a brief description of the steps taken, the sample PDF form is filled out with the data from the sample spreadsheet file.

Home

After starting the program, the "home page" of the program opens. This offers a short welcome text, a reference to this documentation, the

BulkPDF Business [DEMO]	J						1 0		-	ø ×
	Startseite									
	BulkPDF E	Business								
	Vielen Dank, dass In den nächsten Sc Klicken Sie nun bit	Sie sich für die Nutzun :hritten wird Sie das Pro ite, um zu dem nächste	g von BulkPDF Business e ogramm zur Auswahl ein en Schritt zu gelangen, au	ntschieden haben. Die s er Datenquelle, der Zuw f die Schaltfläche unten	Software, welche automa veisung der Daten zu PDF n in der rechten Ecke.	tisiertes Ausfüllen von PE -Formular Felder, der Dat	F-Formularen benutzer einamenseinstellungen	freundlich und datenschutzge sowie der Ausgabe Ihrer PDF-I	echt für jeden Verfügbar Formular-Dateien führen.	macht.
	Bitte nehmen Sie s Klicken Sie hier, un	ich die Zeit einmal in d n die Dokumentation z	den FAQ (häufig gestellte zu öffnen.	Fragen) und in die Dok	umentation hineinzuscha	uen.				
BulkPDF Business	Für einen schneller Klicken Sie hier, um	n Überblick über das Pr n die Beispiel-Konfigur	rogramm steht auch Beis ation zu laden.	piel-Konfiguration zur V	/erfügung.					
PDF	Falls Sie Kritik, Wü	nsche an die weitere Er	ntwicklungen oder allgen	veines Feedback gegebe	en möchten, <u>klicken Sie l</u>	vitte hier.				
2021.12.30.811										
	Software-Lizenz					Software-Version				
	Software-Lizenz-Code:		Lizenz akt	tivieren		Installiert:	2021.12.30.811	Verfügbar:	2021.12.30.0811	
	Lizenz-Status: 🗙 🕅	Verfallsdatum:				Versionskanal: Stabil *		Auf neue Version aktualisieren		
Dokumentation 🖸	Optionen									
Rechtliches 🐡	Sprache: Deutsch : de									•
Copyright © 2021 Julien Bartels	Support 🖅			Kon	nfiguration: Laden R 🗐 S	peichern 3				Weiter →

Clicking the "Next" button in the bottom right corner takes you to the "Data Source" page.

Data source

In this step, the data source type "Excel" is first selected and then the Excel file itself.

Loading the example configuration and a link to the feedback page.





Once the Excel file has been loaded, you can select one of the tables it contains. A preview of the data in that table will then be displayed.

Only the data that is in the correct format for BulkPDF and was found is displayed.

Clicking on "Next" again takes you to the page for the "PDF Form".

PDF form

On this page you must first select a PDF form file.



After a PDF file has been selected, a collection of general data about the file, a PDF preview and the table for assigning the data from the "data source" to the individual fields of the PDF form appear.

	PDF-Datelauswahl							
	PDF-Datei: C\Program Files\BulkPDF Business\ProcessExample.pdf							Auswähle
	Daten-Vorschau							
	Anzahl gefundener nutzbarer PDF-Formular-Felder: 10 Dokumententyp: AcroForm	Erstellungsdatum: 01.12.202	1 14:14:19		Änderungsdatur	m: 03.12.20	21 19:14:42	
	PDF-Vorschau	Zuweisung PDF-Formular-Feld						
	Automatische Aktualisierung: 🗹 🛛 Aktualisieren 🗊 🛛 In System PDF-Viewer öffnen 🔍	Feld	Feldtyp	Aktueller Wert	Daten bearbeiten	Bearbeitet	Schreibschutz	Ausgewählte Daten
	Formularfeld hervorheben:	Vorname	TextBox		Bearbeiten 🕅			
		Nachname	TextBox		Bearbeiten 😭			
		Strasse	TextBox		Bearbeiten 🕅			
BulkPDF Business	る- 参 正 Aa 皇- 女 📐 🌡 🛱 🔍 Q	Hausnummer	TextBox		Bearbeiten 🖹			
		Postleitzahl	TextBox		Bearbeiten 🖹			
		Ort	TextBox		Bearbeiten 📄			
PDF		Geschlecht	RadioButton		Bearbeiten 🕅			
	Beispiel-Bestell-Formular	UmweltfreundlicheVerpackun	CheckBox	False	Bearbeiten 🕅			
		ZugehoerigeBranche	ListBox		Bearbeiten 🕅			
-		DerzeitigeTaetigkeit	ComboBox		Bearbeiten 📄			
	[Company Name] [Street Address, City, ST ZIP Code] [Phone] [Email] [Website]	and the second se						
	TEXTFELD-BEISPIEL (TEXTBOX):							
	TEXTFELD-BEISPIEL (TEXTBOX): Vorname Nachname	Zusätzliche Spezialfelder						* Hinzufügen

Here, the "Last Name" field was opened using the "Edit" button and the data was assigned to the "Last Name (Textbox)" column from the data source.

	Por Daringente De POR Farmular - Optimer -					
	PDP Date: Criticipan Nectbald/Of Business/Proceedbarryin.pdf					Autorities.
	Dates Verschau					
	Anzahl gehandener nutzbarer PDF Formular Folder: 12 Dokumenter	ntyp. AcroForm Enterlangulation: 01.12.2021 14.14.19		Anterungulation	05.12.2021 19.1642	
	POF Verschas					
	Datenzuweisung			Daten bearbeiten	Bearbeitet Scheelkschutz	Ausgewählte Dates
	Feld: Nachname	Feldtyp: TextBax	*	Searbeiten 📴	0 0	
	Schreibschutz:			Bearbeiten 😰	0 0	
	Datenquelle			Bearbeiten 📴		
BulkPDF Business				Bearbeiter 🛐	0 0	
	Datenquelle nutzen: Nachname (Textbox)		*	Bearbeiter 🔮		
	O Konstanter Text:		0	Bearbeiten 😭	0 0	
PDF				Bearbeiten 📴		
	Beispiel-] Schriftart			Baarbarbar 22	0 0	
	the second s			Baarbaiten 🛐		
				Bearbarten 22	0 0	
ALC: NOT	[Company Name] [Sweet Address, City, S [Please] [Email] [Website] TEXTFELD-DESIPTEL (T					
			*			
	Vorsame		0k 🖌			
·	Strafe atthe	Feldige Sale	Post	n Beach	elbung Daten bea	* Hinzufügen deiten Daten bearbeiten
Dekonstation 🕃	Postleitzahl ind					
and a state later	Support (22)	Redgester Later () Spectres ()				- Zarick Weiter -

After closing the editing window, the preview now shows that the first row of the data source was used for demonstration purposes to fill out the form as an example and that "Last Name" now contains the value of the first row of the "Last Name (Textbox)" column from the data source.



Subsequently, all other fields were also assigned the appropriate columns as data

1 BulkPDF Business [DEMO]	sources.									- 0 >
	Startseite Datenqueile PDF-Formular Optionen F									
	PDF-Dateiauswahl PDF-Dateit CAProgram Files\BulkPDF Business\ProcessExample.pdf									Auswählen.
	Daten-Vorschau	Daten-Vorschau								
	Anzahl gefundener nutzbarer PDF-Formular-Felder: 10 Dokumententyp:	AcroForm		Erstellungsdatum: 01.12.202	1 14:14:19		Änderungsdatu	m: 03.12.202	1 19:14:42	
	PDF-Vorschau			Zuweisung PDF-Formular-Feld	6					
	Automatische Aktualisierung: 🗹 🛛 Aktualisieren 🔯 👘 System PDF-Viewer öffne	ien Q		Feld	Feldtyp	Aktueller Wert	Daten bearbeiten	Bearbeitet	Schreibschutz	Ausgewählte Daten
	Formularfeld hervorheben:			Vorname	TextBox		Bearbeiten 🖹	1		Vorname
			1	Nachname	TextBox		Bearbeiten 🕅	1		Nachname (Textbox)
		2 2- 1- +-		Strasse	TextBox		Bearbeiten 🕅	1		Straße
BulkPDF Business	る- 参, 団 Aa 皇- 成 📘 🎍 🛱 🔍			Hausnummer	TextBox		Bearbeiten 🗊	1		Hausnummer
			1	Postleitzahl	TextBox		Bearbeiten 🖹	1		Postleitzahl
	TEXTFELD-BEISPIEL (TEXTBOX):			Ort	TextBox		Bearbeiten 🖹	1		Ort
PDF	Vorname Servit Nathume Herter			Geschlecht	RadioButton		Bearbeiten 🕅	1		Geschlecht (Radio Button /
				UmweltfreundlicheVerpackun	CheckBox	False	Bearbeiten 🕅	1		Umweltfreundliche Verpaci
	Straße Weit Weit Weit			ZugehoerigeBranche	ListBox		Bearbeiten 🖹	1		Zugehörige Branche (Listbi
<u> </u>				DerzeitigeTaetigkeit	ComboBox		Bearbeiten 🖹	1		Derzeitige Tätigkeit (Comb
	Postfeitzahl									
	OPTIONSEELOREESPIEURADIOBOXUSINESS DEMO		:							
	‱ക്ക്മാനഹണാ⊚re ഖികBulk®BF.de)									
	MARKIERFELD (CHECKBOX):									
	Umweltfreundliche Verpackung: 🗹									
	LISTENFELD (LISTBOX):									
	Zuzehörize Branche:			Zusätzliche Spezialfelder						
	Las interior las interior									 Hinzufügen
	KOMBINATIONSFELD (COMBOBOX):			Feldtyp	Seite	Position	Besch	reibung	Daten bea	rbeiten Daten bearbeiten
	Derzeitige Tätigkeit:									
Dokumentation		-								
Rechtliches 🖶										
	Support 🗊	Konfiguration: Laden 지원	Speid	chern J						← Zurück Weiter →

After the data has been assigned to the fields, the final setting step can be reached by clicking on the "Next" button.

RulkPDF Rusiness IDFM01

Options

In this step, the file name can be composed by including formatting elements such as data from a column in the data source, date/time, etc.

1

to the file name generator. Whenever you change these formatting elements, you will immediately see in the preview below, based on the first three file names, what exactly the change will do.

		Formular Optionen =						
	Dateiname				Datainaman Formationung Flom			
	Formatierungs-Element	Daten	Daten-Vorschau		Dateinamen-Formatierungs-Eleme	nte		
	A V A Zeitennummer	2			Konstante:			Hinzufügen
	Konstante -	- inde	pdt		Datenqueller			Hinzufilgen
								inizologen
					Zeilennummer: Hinzufügen	Datum / Uhrzeit: dd.MM.yyyy	Hinzufüg	en i
	Dateinamen-Vorschau 1: 1.pdf							
BulkPDF Business	Dateinamen-Vorschau 2: 2.pdf		Dateiname	n-Vorschau	3: 3.pdf		Liste aller weitere	n Dateinamen
	Ausgabe							
PDE	Ausgabeordner:							Auswählen
100 C								
2021.12.30.811								
	Optionale Optionen							
Dolumentation P	Globaler Schreibschutz: Datei-Kompression:	Zeilen zu Dokument (Beispiel:1,4-9,34):						
	Schriftart				Y Signatur			
Rechtliches 🎡								
Convright © 2021 Julien Bartels	Support 🖾		Konfiguration: Laden 저열	Speich	ern x12		- Zurück	Weiter -

In addition to a good looking file name, the output directory was also selected in which all the files created in the same way will be located.

BulkPDF Business [DEMO]									- (o ×
			PDF-Formular = Optionen							
	Dateiname	Formatierungs-Element	Daten	Daten-Vorschau		Dateinamen-Formatierungs-Elemente				
	A V X	Datenquelle	Nachname (Textbox)	Mueller		Konstante:			Hinzut	ifügen
	* * ×	Konstante	-	<u>2</u> 7						
	* * X	Datenquelle	Vorname	Bernd	<	Datenquelle:			* Hinzuf	fügen
	A V X	Konstante	.pdf	.pdf		7.damente March	Sales (Barra HAMIsee)		Trans Brannes	
						Zestennummer: Hinzurugen	Datum / Unizen: dd.MM.yyyy		tinzutugen	0
	Dateinamen-Vor	rschau 1: Mueller_Bernd.pdf								
BulkPDF Business	Dateinamen-Vor	rschau 2: Mustermann_Max.pdf		Datein	amen-Vorschau	3: Musterfrau_Marie.pdf		Liste alle	r weiteren Dat	teinamen
	Ausgabe									
PDE	Ausgabeordner:	C//Users/tadel/Downloads							Aus	swählen
2021 12 20 911										
2021.12-30.011										
	Optionale Optio	nen								
Deloursestation	Globaler Schreib	oschutz: Datei-Kompression:	Zeilen zu Dokument (Beispiel:1,4-9,34):							
	Schriftan	t				Signatur				
Rechtliches 🔮										
Carriely © 2021 Line Party	Support 🖾			Konfiguration: Laden K	Speiche	m JB		- Zu	rück 1	Weiter
Copyright © 2021/Julien Bartes										

Click on "Next" to go to the final step.

Completion

BulkPDF Business [DEMO]

This step no longer offers any setting options but only serves as an overview of the settings made.

σ

			- Fertigstellung				
	Vorschau			Fertiostellung			
	Ausnahenriner: CillicercitodeRDo	weinade		Zeilennummer	Datei	Fehlerinformationen	Anzeigen
	Ausgabeoruner. Choseisitobeijbo						
	Anzahl gefundener Zeilen: 8	Schreibschutz: Datei-Kon	npression:				
	Anzahl an PDF-Formular-Felder: 1	10 Anzahl an Spezial-Feldern: 0					
	Zeilennummer Datei PDF-Vorschau						
	1 C:\Users\tadel\Dow	nloads\Mueller_Bernd.pdf	□→∞				
	2 C:\Users\tadel\Dow	inloads\Mustermann_Max.pdf	₫→®				
	3 C:\Users\tadel\Dow	inloads\Musterfrau_Marie.pdf	[]→@				
BUIKPDF Business	4 C:\Users\tadel\Dow	mloads\Klein_Berthold.pdf	€→@				
	5 C:\Users\tadel\Dow	nloads\Gro8_Anna-Lisa.pdf	⊡→®				
PDF	6 C:\Users\tadel\Dow	nloads\蛋糕_蛋糕.pdf	□→∞				
	7 C:\Users\tadel\Dow	nloads\topt_topt.pdf	∁→⊚				
	8 C:\Users\tadel\Dow	nloads\i, l, ě, č הקליטה_pdf	[]→@				
2021.12.30.811							
Rechtliches 🚭							
Copyright © 2021 Julien Bartels	Support 🖃		Konfiguration: Laden 🕫 Speic	hem S		- Zurück	Fertigstellen →

By clicking on the "Finish" button at the bottom right, the filling process begins.

BulkPDF Business [DEMO]	Process	S.					- 0 ×
	and the owner of	Delegade - PD famile - Optimer -	Fertigstellung				
	Versilies			Fertipitelung			
	Angeleente	en Cillien/bals/Downloads		Letenummer	Datei	Feblecis/urmationen	Anantigen
	And the state	tere ben d		1	Ciliberritedel Download/Mueller, Bernä pdf		
				2	Chilseni tadel Downloads Mustermann, Max poli		
	Anzahit an POI	F Formulae Felder: 10 Anzahl an Special Feldern: 0		1	Cilisenitadel Dovritadd Muterhau, Maria pdf		
	Zelennammer	Datei	FDF-Verschere	4	Cilibert/tabel Download/ Kain, Berthold pdf		
	1	C/Dan/tabl/Download/Mueler,Bend.pdf	13+ 0		Cillian Cashel Download Clinik Anna Liss of		
	2	C/Diser(table)/Downloads/Mustemann_Mas.pdf	E3+0		Comparing the second		
and the second se	1	Cilliam/tadel/Downloads/MusterNau, Marie pdf	13+0	•	Concernation Designed and a start party of		
BulkPDF Business	4	Citizeni Italel Zouritaelti Kein, Berkuld pdf	12+0	,	C12bert/tade/Downloads/ropt_ropt.pdf		
	1	Ciliberitate/Develoat/Cod. Arra-Lia.off	11+0	4	Cluber/Itale/Dovnload/J. (4, 2 _10/101.pdf		
POF		Property of the second s	il.e.				
	1	Crown table Downloads	8/8				
	-	C'User/tade' Downloadd'	t, (, ě, č مرځان م, pdf				
			Abbrechen				
202112-30.811					lin.		
Datumentation 🖬							
Rectdicion 🖶							
Copyright 6 2021 Julien Barbels	Support SD		Redgestion Later # 20 Spec	(De men		- Zarück	Terliptelles +

After all files have been created, the output folder with the created files opens automatically.

V Heute (11)			
	20 42 2024 42 07		77.1/0
pdf	30.12.2021 12:07	Adobe Acrobat-D	// KB
🔊 торт_торт.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
🙈 Groß_Anna-Lisa.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
🙈 Klein_Berthold.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
👃 Musterfrau_Marie.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
🗹 ᆋ Mustermann_Max.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
🔒 蛋糕_蛋糕.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB
🙆 Mueller_Bernd.pdf	30.12.2021 12:07	Adobe Acrobat-D	77 KB

The configuration you have made should be saved at the end so that you can quickly access it again in the future without having to set anything again.



User interface in Details (GUI)

The following describes all the elements in each window in the application. This chapter is not intended as a "step-by-step" guide to application, but as a reference work for unknown or unclear functions within of the program. This chapter can also provide an overview of the functionality of the application in general. To get started with the application itself, please refer to Chapter 2.

3.1 Home

On the home page you will find basic settings, software license related, update functionality and information regarding the program.

3.1.1 Overview



A Page title: In this bar you can find all the pages of the application. The The currently active page is highlighted by a dark blue background. All other pages are displayed with a light blue background. (The Bar can also be used for quick navigation by clicking on the individual page title is clicked.)

B License: In this element group you can find the current status of the license, its expiration date and a button to activate a license (see The status of the license is shown with a check mark after a successful license activation and verification. If a cross is displayed instead of of the check mark is displayed, no license is active or no valid license is available. the.

C Software version: This is the currently installed software version, the latest available version and a button to update the software. There is also a button for selecting the update channel. In The "Stable" channel contains the latest tested version, while the "Beta" channel is the latest version, which may contain unknown Contains bugs but offers the latest features. **D Options:** This element group currently only offers the selection of the desired program language.

E Page panel: This is where the documentation window (see chapter 3.7.2) as well as legal information, such as software license texts, can be accessed.

F Support: This button opens the support window (see chapter 3.7.1).

G Configuration: In the element group, the current configuration (everything that has been loaded, set and changed since the beginning of the program session) can be saved or an old, already created configuration can be reloaded. (See 3.1.3 for more information)

H Navigation buttons: This area contains buttons for moving forward or backward one page in the program. In the last step, the "Next" button becomes "Finish", which finally creates the PDF files (see completion page 3.5).

3.1.2 License activation

BulkPDF Business [DEMO]						-	o ×
	Starbeite Distanguide MCF France	An - Option - Park					
	BulkPDF Business						
	Vielen Dank, dass Sie sich für die Nutzung von Bull In den nächsten Schritten wird Sie das Programm z Klicken Sie nun bitte, um zu dem nächsten Schritt z	PDF Business entschieden haben. Di ur Auswahl einer Datenquelle, der Zu zu gelangen, auf die Schaltfläche unte	e Software, welche automatisiertes Ausfüllen weisung der Daten zu PDF-Formular Felder, en in der rechten Ecke.	von PDF-Formularen benutze der Dateinamenseinstellungen	freundlich und datenschutzgen sowie der Ausgabe Ihrer PDF-F	echt für jeden Verfügbar ma ormular-Dateien führen.	cht.
	Bitte nehmen Sie sich die Zeit einmal in den FAQ (h Klicken Sie hier, um die Dokumentation zu öffnen.	aufig gestellte Fragen) und in die Do	kumentation hineinzuschauen.				
BulkPDF Business	Für einen schnellen Überblick über das Programm Klicken Sie hier, um die Beispiel-Konfiguration zu la	steht auch Beispiel-Konfiguration zur aden.	Verfügung.				
ACI O MAR	Falls Sie Kritik, Wünsche an die weitere Entwicklun	offware Lizen: Coffware-Lizen: Code Influence-Lizen: Code Vertallodature: B Abbrechen	Oberp	rolen			
	Software Come		Software Venion				
	Seftware Lizenz-Coder	Libert aktivieren	Installert	- 2021 12.00.011	Verligher	2021-12.30.0811	
- P	Linex-Status: X Variabilitation		Vericestanat Stat	2) (Auf raue Version althalformer		
Determination E	Optionen						
Restlichen 🖶	Spracher Deutsch-de						
Copyright & 2027 Julier Bartels	Support 53		entgansteen Lader F 🔄 Specture 10				Welter -

A Software license: The software license code is entered here (e.g. F0-OB-EX-55). After entering the code, the "Check" button must be pressed so that the validity can be checked by the license server and the expiration date of the program key can be displayed.

B License information: This is where you will find the information about the program key that was loaded using the "Check" button.

C Activate license: If the license exists, is valid and has not yet expired, this button can be pressed to create an encrypted license file on the system, which from now on verifies, even offline, that all paid functions within the program can be used.

BulkPDF Business								- 0	×
BulkPDF Business	Securit BulkPDF Business	Tree Tree	Anderungudirum Tvr 06.11.2021 14:03 But	v b P 1 Gode MOT_Configur 43 KB	F Business (*BußPDf)				
	Software Lane			Software Berston					
	Software Lizenz Code	Lore attorne		Installer's	3521.11.16.1757	Verligher	2021.11.96.1757		
	Larre Satur V Verbildeten	25 16/2012 14 1/2 50		Versionskanati Statit *					
Datumentation 🖬	Optionen								
Rectelates 🖤	Sprache: Deutsch - de								1
Capyright © 2021 Auton Barters	Support III		Konfiguration Labor # 20 5	picters x22				Water	-

3.1.3 Configuration: Save & Load

A A previously saved BulkPDF configuration.

- **B** File name and file extension of the BulkPDF configuration.
- **C** The current BulkPDF configuration can be saved here.
Encryption If passwords for signatures or data connections are to be saved with the configuration, this configuration should be saved encrypted. For this purpose, the file extension ".BulkPDFEncrypted" can be selected when saving a configuration, which then creates the file encrypted after entering a password. When loading a ".BulkPDFEncrypted" configuration, the password is requested again to load it.

3.2 Data sources

In this step, you will be asked to select a data source, such as Excel, Microsoft SQL Server, etc. Each data source type has its own setting options and points that must be observed. This section and the individual data source types are discussed below.

3.2.1 Overview



A This button allows you to select one of the currently supported data source types, such as Excel, Microsoft SQL Server, etc.

3.2.2 Excel / spreadsheet files



A A spreadsheet file can be loaded here. For further information on which file formats are supported, see chapter 1.3.2.

B Here one of the tables contained in the spreadsheet file must be selected.

Only data from one table can be selected in a

spreadsheet file can be used simultaneously.



A Spreadsheet: You can see a selected xlsx file and the selected table "Table1".

B Data preview: The number of rows and columns found are displayed in the element group. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loading correctly.

C Spreadsheet data preview: The first 1000

recognized rows of the spreadsheet file are displayed to check whether the file was loaded as intended and the data was recognized.

3.2.3 Microsoft SQL Server



A Connection string: The "Connection string" is displayed here with the example "Server=SERVERADDRESS;Database=BulkPDFBusinessTest;Trusted_Connection=True;". For this data source, the "Database" section in the "Connection string" is important because only the tables from this selected database are listed. the.

B Selection of the table from the database selected in the "Connection string". Only data from one table can be used at a time in the program with this data source.

BulkPDF Business								-	o ×		
			Datenquelle								
	Datenquelle	MicrosoftSQL	Server						~		
	Connection										
	Connections String: Severs ;Database=Buil/90/FausinessTest;Trusted_Connections:True; Verbindungsman										
1											
(4	A en-Vorsch	nau						_			
	Anzahl gefu	undener Zeilen:	: 8				Anzahl gefundener Spalten: 6				
	Nutzbare Sp	palten: Last_na	ame						٣		
BulkPDF Business	Bellenkalku	ulation-Datei-V	Vorschau						$ \longrightarrow $		
	1: Lastname	2: Firstname Barnd	3: Adress	4: CheckboxValue	5: Radiobutton_value	6: ComboboxValue					
	Mustermann	Max	Mustermann Straße	No	2	Test					
PDF	Musterfrau	Marie	Mustermann Straße	Yes	3	Keks					
	Klein	Berthold	Kleiner Weg								
(<u>au</u>	Groß	Anna-Lisa		No	4						
	蛋糕	蛋糕	печенье	No	4						
	торт	торт	饼干	Yes	3	Keks					
2021.11.16.1757	ř, ľ, ě, č	ř, ľ, ě, č	ř, ľ, ě, č								
Dokumentation											
Rechtliches 🐏	2										
Copyright © 2021 Julien Bartels	adpport Exil						Kontiguration: Laden N Speichern M	- Zurück	Weiter		

A You can see the "Connection string" and the selected table "Example".

B The number of rows and columns found are displayed in the element group. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loaded correctly.

C At this point, the first 1000 recognized rows of the SQL table are displayed in order to check whether it was loaded as desired and the data was recognized.

BulkPDF Business							o ×
	Startseite Datenquelle =	PDF-Formular Optionen Fertigs					
	Datenquellen-Typ: Benutzerdefinierte Microsoft SQL-	Abfrage					•
	Verbindung						
	Onnection String:					Verbindungsmana	iger öffnen.
	QL Query:		4				a V
	B)		Austuhren				
BulkPDF Business			1				
PDF	Geben si	e eine Connection-String	sowie eine SQL-Qu	ery und drücken folge	nd auf "Ausführen".	7	
لا غد	Hinweise zum Datenquellen-Typ						
2022 04 05 0807							
LULLON 05 1000							
		Für den die Nutzung von Daten von einem Microsoft SQL Serve	er stehen mehrere Varianten zur Datenbeschaffung	zur Verfügung.			
		Zum einen kann die Option gewählt werden, bei der man schlik Von diesen Tabellen kann dann eine ausgewählt werden, die als Für für gültige "Connection-String"-Beispiele schauen Sie sich I	cht einen "Connection-String" eingibt und dann für s Datenquelle fungiert. bitte die Dokumenation an.	die jeweilige Datenbank automatisch alle verfügbaren Tab	ellen angezeigt werden.		
		Die zweite Option ist die Möglichkeit, neben dem "Connection Diese Option erlauht damit die volle Funktionalität von T-SOL a	-String", schlicht eine SQL-Abfrage einzugeben, die auf dem Microsoft SQL Server für die Datenquelle in	einem die gewünschten Daten ausgibt. nerhalb von BulkPDE zu nutzen			
Dokumentation							
Rechtliches 🚭							
Convright © 2022 Julien Bartele	Support 🖾	Kor	nfiguration: Laden 또 앱 Speichern 뇌앱			← Zurück	Weiter -
and a concertation of the							

3.2.4 Microsoft SQL Server Custom Query

A The "Connection string" is specified here.

BulkPDF Business	B The	SQL	que	ry is wr	itten h	here.			- a ×				
	Startseite	- Date	enquelle	- PDF-Form									
	Datenquellen-Typ	Benutzerdefi	inierte Micros	oft SQL-Abfrage					×.				
	Verbindung												
1	enerction String Server- Oatabase=BulkOPBusines/Text Trutted Connection True												
	A SQL Query: SE												
1	Austilven												
BulkPDF Business	Anzahl gefunder Nutzbare Spalter	ner Zeilen: 8 n: Fullname						Anzahl gefundener Spalten: 7	•				
PDF	C)	ion-Datei-Vors	chau										
	1: Fullname	2: Lastname	3: Firstname	e 4: Adress	5: CheckboxVa	alue 6: Radiobutton	value 7: ComboboxValu	e					
	Bernd Mueller	Mueller	Bernd	Weit-Weit-Weg	Yes	1	Kuchen						
	Max Mustermann	Mustermann	Max	Mustermann Straße	No	2	Test						
2022 04 05 0807	Berthold Klein	Klein	Berthold	Kleiner Weg	ies		Nets						
LULLOHOTOUT	Anna-Lisa Groß	Groß	Anna-Lisa		No	4							
	蛋糕 蛋糕	蛋糕	蛋糕	печенье	No	4							
	торт торт	торт	торт	饼干	Yes	3	Keks						
	f, [, é, č f, [, é, č	7, 1, ê, č	ř, ľ, ě, č	ř, ľ, ě, č									
Dokumentation 🖱													
Copyright © 2022 Julien Bartels	Support E						Konfiguration	: Laden K 🗃 Speichem 뇌열	- Zurück Weiter -				

41

A Connection string: The "Connection string" is displayed here with the example "Server=SERVERADDRESS;Database=BulkPDFBusinessTest;Trusted_Connection=True;". For this data source, the "Database" section in the "Connection string" is important because only the tables from this selected database are listed.

B Here you can see the SQL query.

C The number of rows and columns found are displayed in the element group. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loaded correctly.

D At this point, the first 1000 recognized lines of the SQL query are displayed to check whether the file was loaded as desired and the data was recognized.

3.2.5	Microsoft	Access
-------	-----------	--------

BulkPDF Business			o ×
	Startselle – Datenquelle – FDF-Formular – Optionen – Fertigstellung		
	Datenquellen-Type Microsoft Access		
			_
	Verbadung		
	A	A	uswählen.
	K Andrew Control of the Control of t	* Ak	tualisieren
(B)		
BulkPDF Business	f f		
PDF	Wählen Sie eine Microsoft Access Datenbankdatei und eine Tabelle aus.		
	Hinweise zum Datenquellen-Typ		
	Allgemeine Informationen		
2022.04.05.0807	Für den die Nutzung von Daten von einem Microsoft Access stehen mehrere Varianten zur Datenbeschaffung zur Verflügung.		
	Zum einen kann die Option gewählt werden, bei der man schlicht eine Datenbank-Datei (accdb oder .mdb) auswählt und dann für die jeweilige Datenbank automatisch alle verfügbaren Tabellen angezeigt werden. Von diesem Tabellen kann dann eine ausgewählt werden, die als Datenquelle fungiert.		
	Die zweite Option ist die Möglichkeit, neben der Datei, eine SQL-Abfrage einzugebeit, die einem die gewönschten Daten ausgibit. Diese Option erlaubt damit die volle Funktionalität von SQL auf der Microsoft Access Datenbank für die Datenquelle innerhalb von BulkPDF zu nutzen.		
	Haufig gestellte Fragen (FAQ)		
	System.InvalidOperationException: The 'Microsoft.ACE.OLEDB.12.0' provider is not registered on the local machine.		
Dokumentation			
Rechtliches 💮			
Copyright © 2022 Julien Bartels	Support 🗊 Konfiguration: Laden 🕫 Speichern 🗐	- Zurück	Weiter

A A Microsoft Access database file can be loaded here.

B Selecting the table from the selected Microsoft Access database file.

Only data from one table can be used at a time in the program with this data source.

A You can see the selected Microsoft Access database file and the selected table "Table 1 (customer data)".

B The element group displays the number of rows and columns found. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loading correctly.

C At this point, the first 1000 recognized rows of the table are displayed to check whether it was loaded as desired and the data was recognized.

	3.2.6 Microsoft Access Custom Query		
BulkPDF Business			σ
	Startseite – Datenquelle – PDF-Formular – Optionen – Fertigstellung		
	Datenquellen Typs: Microsoft Access Benutzerdefinierte Abfrage		
	Verbindung		
	Autei		Auswähler
	Austilitien		
BulkPDF Business	Wählen Sie eine Microsoft Access Datenbankdatei und führen eine SQL-Query aus. Himeise zun Datengaellen Typ Algemeine Informatione For den de Nutzung von Daten von einem Microsoft Access schem mehrere Varianten zur Datenbaschaffung zur Verfügung. Zum einen kann de Option genächt werden, bei der man schlicht eine Datenbaschaffung zur Verfügung.		
	Von diesen Babellen kann dann eine ausgewählt werden, die als Datenquelle fungiert. Die zweite Option ist die Möglichkeit, neben der Datei, eine SQL-Abfrage einzugeben, die einem die gewünschten Daten ausgibt. Diese Option entsch damit die volle Funktionalität von SQL auf der Microsoft Access Datenbank für die Datenquelle innerhalb von Buli/PDF zu nutzen.		
	System InvalidOperationException: The "Microsoft.ACE.OLEDB.12.0" provider is not registered on the local machine.		
Dokumentation			
Rechtliches 🖶		- Zurich	Wette
Copyright @ 2022 Julien Bartels	apport an Norringaration, Laden N Spectrem 3	- Zuruck	weite

A A Microsoft Access database file can be loaded here.

	Date	tartseite — Da	stenquelle Access Benu	PDF-For									
	-												
	Ver	bindung						1014				1 (aux	
A Meeting Michael actb												AUSV	
	× so	Configure SELECT - FROM [labelea] (Kundendaten)):											
(R)	Austiliten											
DF Business	Ar	vzahl gefundener Zeilen: 8 utzbare Spalten: ID							Anzahl gefundener Spalten: 11				
	()	pellenkalkulation-Datei-Vo	orschau										
- m	1: ID	2: Nachname (Textbox)	3: Vorname	4: Straße	5: Hausnummer	6: Postleitzahl	7: Ort	8: Geschlecht (Radio Button / Radiobox)	9: Umweltfreundliche Verpackung (Checkbox)	10: Zugehörige Branche (Listbox)	11: Derzeitige Tätigkeit (Combobox)		
	1	Mueller	Bernd	Weit-Weit-Weg	23	10178	Berlin	1	1	п	Angesteller		
<u> 264</u>	2	Mustermann	Max	Mustermann Straße	5	80331	München	1	0	Industrial	Ehrenamtlicher		
	3	Musterfrau	Marie	Mustermann Straße	7	31515	Wunstorf	2	Yes	Cake-Factory	Freelancer		
2.04.05.0807	4	Klein	Berthold	Kleiner Weg	3	70173	Stuttgart	3	No	Tax	König von Deutschland		
	5	Groß	Anna-Lisa	Blumenweg	13	40210	Düsseldorf	3	Ja	Top secret	Ehrenamtlicher		
		重糕	蛋糕	快干	17	34117	Kassel	2	A.	Industrial	Ehrenamtlicher		
	6			Develope	2	39104	Magdeburg	1	нет	Cake-Factory	Angesteller		
	6 7	торт	торт	ine nembe						177	Annostallar		
	6 7 8	торт ř, ľ, ě, č	דסףד הקליטה	เงอิน	37	24103	Ormet	1	si		Angesteller		
	6 7 8	τορτ , (, (, , , , ζ	торт пофо	hga	37	24103	iðanş	1	8	11	Algebied		

A The selected Microsoft Access database file is visible.

B Here you can see the SQL query.

C The number of rows and columns found are displayed in the element group. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loaded correctly.

D At this point, the first 1000 recognized lines of the SQL query are displayed to check whether the file was loaded as desired and the data were recognized.

BulkPDF Business						-	U X
	Sta	seite – Datenquelle – PDF-Formular		gstellung			
	Daten	ellen-Typ: JSON					
	-						
	Tabel	nkalkulation					
	Date					1	Auswählen.
(A)-						
BulkPDF Business				1			
				Wählen Sie eine Datei a	115		
				Wallell Sie eine Datel a	us.		
PDF							
		Hinweise zum Datenquellen-Typ					
2022.04.05.0807							
				In Entwicklung-			
Dokumentation							
Rechtliches							
	Support [x	ionfiguration: Laden 또 🗋 Speichern 고말		← Zurück	Weiter
Copyright © 2022 Julien Bartels							

3.2.7 **JSON**

A A JSON file can be loaded here.

Dr business	_							- (
	Startseite		Datenquelle					
	_							
	Datenquellen	-Typ: JSON						
	Tabellenkalk	ulation						
	Datei:				ور	ON\DocExamp	Lison	Ausy
(hau						
	Anzahl gefu	ndener Zeilen:	: 8			A	ahl gefundener Spalten: 6	Änderungsdatum: 05.04.2022 13:06:43
	Nutzbare St	alten: Last na	ame					
IKPDF Business	B)	latin Data	Management					
	1: Lastname	2: Firstname	3: Adress	4: CheckboxValue	5: Radiobutton value	fr Comhohov)	hae	
PDF	Mueller	Bernd	Weit-Weit-Weg	Yes	2	Kuchen	101	
	Mustermann	Max	Mustermann Straße	No	1	Test		
	Musterfrau	Marie	Mustermann Straße	Yes	3	Keks		
<u> </u>	Klein	Berthold	Kleiner Weg					
	Groß	Anna-Lisa		No	4			
022 04 05 0907	蛋糕	蛋糕	печенье	No	4			
122.04.03.0001	торт	торт	饼干	Yes	3	Keks		
	ř, ľ, ě, č	ř, ľ, ě, č	ř, ľ, é, č					
Dokumentation 🖸								
Rechtliches 🖶	$\hat{\mathbf{a}}$							

A File: Shows a selected JSON file.'

B Data preview: The number of rows and columns found are displayed in the element group. A list of the columns found, which can be used in the next steps for data selection, is also shown for an overview and to check that the file is loading correctly.

C Data preview: The first 1000 recognized lines of the

file to check whether the file was loaded as intended and the data was recognized.

3.2.8 XML

BulkPDF Business			– ø ×
	Startseite Datenquelle PDF-Formular Optionen		
	Datenquellen-Typ: XML		•
	Tabellenkalkulation Date:		Auswählen.
(A)	↑	
BulkPDF Business		Wählen Sie eine Datei aus.	
PDF	Hinweise zum Datenquellen-Typ		
2022 04 05 0807			
		In Entwicklung.	
Dokumentation			
Rechtliches 🖶	Survey 70	Valuation Inter 70 Carbon 10	. Zwitch Walter -
Copyright © 2022 Julien Bartels	addbourran	konngarauon, cauen n g speicnem a g	- zuruck Weiter -

A An XML file can be loaded here.

BulkPDF Business									-	o ×								
	Startseite		Datenquelle															
	Datenquellen	-Typ: XML								×								
	_																	
	Tabellenkalk	culation																
	Datei:				V	ML\Example	xml			Auswählen								
	A)ten-Vorre	hau																
<u></u>	Anzahl gefu	indener Zeilen	: 8				Anzahl gefundener Spalten: 6	Änderungsdatum: 31.03.2022 07:52:41										
BulkPDF Business	Nutzbare Sp	palten: Last_ni	ame							+								
Durki Di Dusiness	B)	ulation Datei	Vorchau							$ \longrightarrow$								
	1: Lastname	2: Firstname	3: Adress	4: CheckboxValue	5: Radiobutton value	6: Combob	xValue											
PDF	Mueller	Bernd	Weit-Weit-Weg	Yes	2	Kuchen												
	Mustermann	Max	Mustermann Straße	No	1	Test												
	Musterfrau	Marie	Mustermann Straße	Yes	3	Keks												
	Klein	Berthold	Kleiner Weg															
	Groß	Anna-Lisa		No	4													
2022.04.05.0807	蛋糕	蛋糕	печенье	No	4													
	торт	торт	饼干	Yes	3	Keks												
	ł, l, ė, č	ř, l, ê, č	ř, ľ, é, č															
Dokumentation																		
	Ц.																	
Rechtliches 🔮	C)																	
Copyright © 2022 Julien Bartels	pport 📰						Konfiguration: Laden 또 앱 Speichern 뇌앱		+ Zurück	Weiter								



B Data preview: In the element group, the number of found rows and columns found are displayed. A list of the found Columns that can be used in the next steps for data selection are shown for an overview and to check that the file is loading correctly.

C Data preview: The first 1000 recognized lines of the File is displayed to check whether the file was loaded as desired and the data was recognized.

3.2.9 Spreadsheet / Excel (Swapped columns and rows)

This data source type behaves similarly to the normal "spreadsheet / Excel" data source types (see section 3.2.2) with the difference that Loading the columns and rows are swapped. This behavior can be used, for example, A spreadsheet file must be loaded that has the following structure:

First name E	Bernd	Max	Mary	
Last name M	lueller	Mustermann	Model woman	
Address	Far-Far-Away Mus	stermann Street Musterr	mann Street	

3.2.10 MySQL/MariaDB Server



A Connection string: At this point the "Connection string" with the example ,Server=SERVER ADDRESS;Database=BulkPDFBusinessTest;User Id=myUsername;Password=myPass The "Database" section in the "Connection string" is of particular importance, as only the tables of the specified database are listed.

B Here the selection of the table from the specified in the "Connection string" takes place. database. Please note that only data from a single table can be used at the same time within this data source.

	Startseite Datenquellen Verbindung Connection	-Typ: MySQL String: Serve	Datenquelle	swahl BulkPDFBusinessTest	lar Verbl	eibende Optionen	Fertigstellung	Verbinden
BulkPDF Business	Daten-Vorsci Anzahl gefu Befundene r	ample hau indener Zeilen nutzbare Spalt	e 8 <mark>ten:</mark> Last_name				Anzahl gefundener Spatten: 6	· Altuskueren
PDF	Tabellenkalk	ulation-Datei	-Vorschau					
	1: Lastname	2: Firstname	3: Adress	4: CheckboxValue	5: Radiobutton_value	6: ComboboxValue		
	Mueller	Bernd	Weit-Weit-Weg	Yes	1	Kuchen		
	Mustermann	Max	Mustermann Straße	No	2	Test		
	Musterfrau	Marie	Mustermann Straße	Yes	3	Keks		
2023 06 26 0946	Klein	Berthold	Kleiner Weg					
2023.00.20.0340	Groß	Anna-Lisa		No	4			
	蛋糕	蛋糕	печенье	No	4			
	торт	торт	饼干	Yes	3	Keks		
	ř, ľ, ě, č	ř, ľ, ě, č	ř, ľ, ě, č					
Dokumentation 🗎								
Copyright © 2023 Tadelsucht UG (haftungsbeschränkt)	pport 🖾						Konfiguration: Laden R Speichern J	k Weiter –

A Here you can see the entered "Connection string" and the selected table "Ex-ample".

B This area shows the number of identified rows and columns. In addition, a list of the recognized columns is provided, which can be used in the next steps for data selection.

C This section displays the first 1000 rows of the SQL table. This allows you to check whether the table has been loaded correctly and the Data was adequately recognized.

BulkPDF Business		– ø ×
	Startseite Datenquelle PDF-Formular Verbleibende Optionen Fertigstellung	
	Datenquellen-Typ: MySQL / MariaDB Benutzerdefinierte Abfrage	•
	Verbindung	
(A primetion String: Server=localhostDatabase=8ulkPDFBusinessTestUser ID=rootPassword=	
	SQL Query: SELECT * FROM example	÷
BulkPDF Business	B f Geben sie eine Connection-String sowie eine SQL-Query und drücken folgend auf "Ausführen".	
	Hinwelse zum Datenquellen Typ	
2023.06.26.0946	För die Verwendung von Daten aus einem MySQL- oder MariuD8-Server stehen mehrere Varianten zum Datenabruf zur Verfügung. Zum einen kann die Option gewählt werden, bei der man einfach einen "Connection String" eingbit und dann automatisch alle verfügbaren Tabellen der jeweiligen Datenbank angezeigt werden. Aus diesen Tabellen kann dann eine ausgewählt werden, die als Datenquielle fungieren roll. Bespiele für griftige Verbindungsstring finden Sei nich der bonzemtation. Die zweite Option ist die Möglichteit, neben dem Verbindungsstring einfach eine SQL-Abfrage einzugeben, die die gewünschten Daten zurückgibt. Mit dieser Option können Sie die volle Funktionalitat von SQL auf dem Server für die Datenquiele in BulkPDF nutzen.	
Dokumentation 🖺		
Copyright © 2023 Tadelsucht UG (haftungsbeschränkt)	Support [3] Konfiguration: Laden K Speichern J	Weiter -

3.2.11 MySQL/MariaDB Server Custom Query

A The "Connection string" is entered here.

BulkPDF Business			r quei	y 13 10	mulate	uner	7 .	- ø ×
18		-	Datenquelle					
	Datenqueller	n-Typ: MySQL	. / MariaDB Benutzerde	efinierte Abfrage				
	Verbindung)						
	A SQL Query:	SELECT * FR	er=localhost;Database= OM example;	BulkPDFBusinessTest	User ID=root;Password	-		
(B						Ausführen	
BulkPDF Business	Anzahl gefu	chau undener Zeilen nutzbare Spal	ten: Last_name				Anzahl gefundener Spalten: 6	·
	C bellenkall	kulation-Datei	i-Vorschau					
	1: Lastname	2: Firstname	3: Adress	4: CheckboxValue	5: Radiobutton_value	6: ComboboxValue		
	Mueller	Bernd	Weit-Weit-Weg	Yes	1	Kuchen		
	Mustermann	Max	Mustermann Straße	No	2	Test		
2023.06.26.0946	Musterfrau	Marie	Mustermann Straße	Yes	3	Keks		
	Klein	Berthold	Kleiner Weg					
	Groß	Anna-Lisa	Arcana 25270	No	4			
	36.65	家務	печенье	No	•	Kake		
	7747	1001	2122	165	5	NEKS		
Dokumentation 😭 Recttliche Hinweise 😵								
Copyright © 2023 Tadelsucht UG (haftungsbeschränkt)	Support Ed						Konfiguration: Laden 자입 Speichern 뇌업	- Zurück Weiter -

B The SQL query is formulated here

A Connection string: At this point, the "Connection string" can be viewed with the example ,Server=SERVERADDRESS;Database=BulkPDFBusinessTest;User Id=myUsername;Password=myPass. For this data source, the "Database" section in the "Connection string" is important because only the tables from this selected database are listed.

B Here you can see the formulated SQL query.

C This section displays the number of rows and columns found. A list of the recognized columns is also visible, which can be used in subsequent steps when selecting data.

D This area displays the first 1000 lines of the SQL query. This is to check whether the query was loaded as intended and the data has been correctly identified.

3.3 PDF form

In this step, the data from the data source is assigned to the fields of the PDF form. This is supported by a live preview of the PDF with the data from the first line of the data source. Furthermore, special fields such as QR codes, barcodes, hyperlinks, images, etc. can be added to the PDF document.

the.

3.3.1 Overview



A PDF file selection: In this element group, the PDF form to be filled out can be



A This element group shows the "number of usable PDF form fields found", the document type or document format (AcroForm or XFA), the creation date and the modification date. The creation date and modification date are only displayed if they could be determined.

B PDF Preview: The group of elements contains the preview of the PDF form. Any change to the assignment of data to the PDF will be reflected here with the first row of the data source to provide a better understanding of the Changes. Likewise, the Functionality group to highlight form fields that may have been poorly named by the author of the PDF form, so that you can see which field is hidden behind the cryptic name. (For more information on PDF preview, see section 3.3.2.)

C Assignment of PDF form field: This table lists all form fields found in the PDF form. The order depends on the order in which the author last edited the form fields

which in most cases corresponds to the vertical order of the fields in the PDF itself corresponds, but this does not always have to be the case. The individual lines can be the data assignment of the fields can be changed. (For more information, see for this element group see section 3.3.3.)

D Additional special fields: In this element group, special fields such as Barcodes, hyperlinks, images, etc., can be added to the document. How Even with the form fields that are already included in the document, it is possible to dynamically assign data from the data source to the added special fields. (See section 3.3.14 for more information .)

3.3.2 PDF preview

BulkPDF Business				- o ×							
	Startselte Datenquelle PDF-Formular Optionen Fo										
	PDF-Datejauswahl	PDF. Addusted									
	PDF-Date: C\Users\ \PDF\Example.pdf			Auswählen							
	Dater Vorschau										
	Anzahl gefundener nutzbarer PDF-Formular-Felder: 6 Dokumententyp: An	croForm Erstellungsdatum: 07.02.2012 17:47	.09 Ānde	rrungsdatum: 03.07.2015 07:56:34							
	PDF-Vorschau	Zuweisung PDF-Formular-Feld									
	Automatische Aktualisierung: 🖌 🛛 Aktualisieren 🔯 👘 System PDF-Viewer öffner	n 🖓 Feld	Feldtyp Aktueller Wert	Daten bearbeiten Bearbeitet Ausgewählte Daten							
	Formularfeld hervorheben:	textbox1 Te	xtBox	Bearbeiten 🖹 🗹 [#]kuchen							
		checkbox C	heckBox False	Bearbeiten 📄 🗌							
(combobox C	omboBox	Bearbeiten 📴 🗹 Kuchen							
BulkPDF Business	В	listbox Li	stBox	Bearbeiten 🗟 🗌							
	T	textbox2 Te	xtBox	Bearbeiten 📷							
	TEXTBOX 1: kuchen TEXTBOX 2:	radiobutton R	adioButton	Bearbeiten 🖹							
2021.11.22.1638		E Zusätzliche Spezialfelder									
	LISTBOX:	5.14-	C.I.	Perturban Data barbaha Data barbaha							
Dokumentation	BulkPDE Business DEMO	Feddyp	Sette Position	Descriveroung untern Dearbeiten Daten bearbeiten							
Rechtliches @	C)										
Copyright © 2021 Julien Bartels	Support E	Konfiguration: Laden 또 Speichern 고겔		- Zurück Weiter -							

A At this point, the automatic update of the PDF preview can be disabled Changes to the data assignment can be deactivated if this is necessary for reasons of Computer performance appears necessary. Furthermore, the update can be manually the preview or the preview PDF can be opened in the standard PDF viewer stored in Windows (for changing the standard PDF viewer, see section 7.11). The element group also offers a function to open a form field after selecting its name.

field in color. This function is especially useful if the author of the PDF form, the form fields were not clearly named and only names such as "Text Field 1" were given instead of, for example, "Customer Address".

B Here you will find functions for the PDF preview, such as zoom, a search etc.

C This is where the preview PDF itself is located. Any changes to this preview are temporary and will be discarded the next time the preview is updated. If you really want to save the preview, you can do so by clicking the button "Open in system PDF viewer". Depending on the standard PDF viewer stored in your

operating system, you can permanently save the PDF preview on your system.



3.3.3 Assigning data to PDF form fields

A Field: This column displays the names of the form fields.

B Field Type: This column shows the field types of the individual form fields, such as e.g. text fields, checkboxes, radio buttons, combo boxes, list boxes, etc.

C Current Value: Shows the current content of the form field if no Processing is carried out by BulkPDF Business.

D Edit data: This button takes you to the individual configuration windows for the various form fields and their different field types.

E Edited: Shows whether the data assignment in BulkPDF Business has been edited for the form field.

F Selected data: This column is a representation of the assigned Data.

Write protection This column indicates whether write protection has been applied to the field.

Automatic field data assignment Under the normal assignment table there is the button "Automatic field data assignment". This button automatically assigns the appropriate columns from the data source to the fields. It compares how many characters there are between the name of the form field and the individual column name from the data source. The column name with the smallest difference is then taken. The maximum difference that is allowed can be specified in the text field "Maximum name difference for assignment" next to the button. The default value is "2". This means that, for example, the field "Customer" would be assigned the column "Customa", since only two changes are necessary to make "Customer" into "Cus-tom". These changes are removing the "r" at the end of the sentence and replacing the "e" with an "a". (See https://de.wikipedia.org/wiki/Levenshtein-Distanz)

Practical tip from a user: If the automatic assignment with the If the default value 2 worked, try it with 1 and 0. To see which fields may not have been automatically recognized, scroll please go to the end of the field list.



3.3.4 Textbox & General Field Data Selection

A The form field name and the field type, such as text fields, check boxes, radio buttons, combo boxes, list boxes, etc., are displayed here.

B Write protection: This box can be used to write protect the formula rfeld must be activated. This means that this field is deleted after processing can no longer be processed by BulkPDF Business. The data that BulkPDF written in the field are thus fixed and unchangeable. (For an example and for further information see section 3.3.7).

C No data assignment: If this option is used, the form field retains the content it already has.

D Use data source: With this option, a column from the data source The field will then be filled with the value from this column filled. (See section 3.3.5 for more information).

E Constant text: This option allows you to enter a constant text in the form field that is ultimately the same for all completed PDF documents. (See section 3.3.6 for further information).

F Font: Here you can enter the font of the form field (see see section 3.7.4).

🛃 BulkPDF Business	Sartuda Daungada - PDF Farmular - Optimum -	Tripping		- 0 ×
	PDF Datelaamadd			
	PDF Dates, CrUver ampleFile/20Fil.vang	lapd		AutoPier,
	Dates Verschas	to before the states and states		Annual
	POF Version		_	
	Datenzuweisung 🕅		e Wart	Dates beacheiten Beacheitet Ausgewählte Dates
	Feld: textbox1	Feldtyp: TextBox	-	Basteter 🕅 🗌
	Schreiberhutz		-	Beartanian (2)
	Datenquelle		-	Bearbeiten 😰
BulkPOF Business	O Keine:		<u> </u>	Restates 2
	Datenquelle nutzen: Last name		J	Besteter 2
POF	TEXTBOX 1. Konstanter Text:		2	Besteter 2
	(v) Schriftart			
	CHECKBOX			
	61			
2021 11 16 1757	Y//			
	TA			
	AND BUTTON			
	(2000)000			
			12	
			× •	
	1.			- Hinsulager
	LISTROX	Feldyp Sele	Position	Beschveibung Daten bearbeiten Daten bearbeiten
	YO.			
Defensetation [2]				
	5			
Rechtlichen 🖤				
a mala a series of	Support III	Konfiguration: Laten # 23 Spectrem x23		- Zarick Weter -
Copyright & 2021 Julien Bartels				

3.3.5 Use data source for form field

Here you can see that the user is currently in the editing window for the field "textbox1" and the data source is being used for this (see A).

The data source column "Last name" is selected for the data source, which means that the field is filled with the data from this column.



After the editing window has been closed, the PDF preview updates automatically and shows the completed PDF form with the data from the first line of the data source for the changed configuration. This can be seen here marked at **A**. The value or content of the first line of the "Last name" column was filled into the "textbox1" form field, which in this case is "Mueller".

The fact that the form field was manually edited in BulkPDF Business can also be seen in **B**, where the "Edited" column is now marked and a representation of the selected data can be seen.

BulkPDF Business				- 0 ×
	Sarbala Desequile POF Formula	Continues Continues		
	PDI-Datelaureadd			Provide State
	PDF Dates. C'shar	v PCP (sample pdf		Autodrien,
	Datas Varachau Anzahl gehandener mutdiserer PDF-Formulae-Felder: 6	Dokumantentyp: Acoform Entellurgolatum: 07.02.2012 1747-09		derungsdatum: 03.07.2015.07.56.34
	PDI-Verschau			
	Datenzuweisung 📝		e Wert	Daten bearbeiten Bearbeitet Ausgewählte Daten
	Feld: textbox1	Feldtyp: TextBox	· ·	Bastatan 2 M Kuthan
	🚓 👘 🗛 🚨 🗸 Schreibschutz 🗌			Beatlanter (2)
B. B. BOT B. Lance	Datenquelle			(Restarting)
Duikrut Dusiness	O Keine:			(Restance of)
				(Automotion D) ()
POF	Konstanter Text: Ku	hen		[manage] C
	Schrittart			
P	CHEDRIDA			
	01.			
20211116-1757	1/2			
	RADIO BUTTON			
	сомвовок.		6. S	
			0.4	
				- Hondoor
	LISTECK	February 1	laite Position	Bruchenburg Dates bearbeites Dates bearbeites
	6	•		
Dokumentation D		<u>^</u>		
Rechtlichen 🖤	1			
Copyright & 2021 Julien Barters	Support ISI	Konfiguration Laten F(3) [spectrum or(3)]		- Zarisk Weter -

3.3.6 Set constant text for form field

The constant behaves in the same way as "Use data source" (see section 3.3.5) with the difference that the content that is filled into the form field is the same for all documents, regardless of the data source. This can be seen in A, where the value "Cake" was entered for the form field "textbox1". The constant value can already be seen in the background in **B** in the field.

3.3.7 Write protection on individual form fields

BulkPDF Business						- Ø X
	Retails - Descarits - PDF F	ende - Option - Teliptellarg				
	PDF-Datelaumente					
	PDF-Datest C1Userv	sampleFiles/PDP2sample.pdf				Autophies,
	Dates Verschau	1				
	Ancald gehandener nutcharer PCF-Formular-Felder: 1	Dokumantantyp Accilians	Evolutiongostations: 07.01.2012.17.47.09	Ander	ungedatume 03.07.2013 07.56.34	
	Datenzuweis	ung 🗟		Wart	Daten bearbeiten Bearbeitet	ungewählte Daten
	Feld: textbox1	Feldtyp	: TextBox	14	Bastatan (2) 12 19mm	-
	Wreibschutz	7		_	Beatleter 🖹	
Bull DOT Burlinson	(A)			-	Beatater 2	
Duikror Dusiness	Keine:	le nutzen:		-	Besteler B	
					Bastatas (2)	
PDF	R	Text: kuchen				
	CHECKBOR	t				
	RADIO BUTTON					
	COMPORT					
				141		
	LISTBOX			Ok 🖌		
						Himuligen
			Feldyp Sete	Paultan	Bescheelburg Datan bearbe	ten Daten bearbeiten
	BulkPDF Bu	siness DEMO				
Dokumentation 🖬	(Learn more	on BulkPDEde)				
Rachtlaber 🖤	L					
County III 2021 Julies Bartes	Support III	Reefgeration Later 1	() Spectres ()			Surikā Weiter -

In this example you can see that a check mark has been placed in the write protection box (A) and that the write protection has been selected for the form field "textbox1". In **B** you can see that the constant text "kuchen" has been set for the data. became.

BulkPDF Business						- 0 X	
	Startseite – Datenquelle – PDF-Formular – Optionen –						
	PDF-Dateiauswahl						
	PDF-Datei: C\Usen: ExampleFiles\PDP\Example.pdf					Auswählen.,	
	Daten-Vorschau						
	Anzahl gefundener nutzbarer PDF-Formular-Felder: 6 Dokumententyp:	AcroForm En	stellungsdatum: 07.02.2012 17:47:09	Ände	erungsdatum: 03.07.2015 07:56:34		
	PDF-Vorschau	Zuw	eisung PDF-Formular-Feld				
	Automatische Aktualisierung: 🗹 🛛 Aktualisieren 🗊 🗌 In System PDF-Viewer öffr	nen G	Feld Feldtyp	Aktueller Wert	Daten bearbeiten Bearbeitet Ausgew	ählte Daten	
	i∢ ∢ 1 /1 ▶ ▶1 97% ▼ ⊖ ⊖ ⊡ ⊡ № ₩ Ø Q	te	rtbox1 TextBox		Bearbeiten		
		ch	eckbox CheckBox	False	Bearbeiten 🖹 🛛 🖁		
		co	mbobox ComboBo	¢	Bearbeiten		
BulkPDF Business		lis	tbox ListBox		Bearbeiten 😰		
	TEXTBOX 1: kuchen TEXTBOX 2:	te	tbox2 TextBox		Bearbeiten 📄		
	(A)	ra	diobutton RadioButt	on	Bearbeiten 📄		
	снесквох:						
2021 11 16:1757		1					
	LISTBOX	Zi	Zusätzliche Spezialfelder				
						Hinzufügen	
			Feldtyp Seite	Position	Beschreibung Daten bearbeiten D	aten bearbeiten	
	BulkPDE Business DEMO						
Dokumentation	DUIREDI DUSITIESS DEMO						
	(Learn more on BulkPDF	de)					
Rechtliches 😴							
Copyright © 2021 Julien Bartels	Support (22)	Konfiguration: Laden R Speicher			- Zurück	Weiter	
A CONTRACTOR OF A CONTRACT OF							

A Here you can see that the "textbox1" form field can no longer be edited and the content of the field now contains the constant text "cake".

B When representing the selected data you can see the constant text "kuchen".

3.3.8 Date formatting for form fields

To explain date formatting in form fields, a short example is shown below:

Datenquellen-Typ: Tabellenkalkula	tion / Excel
Tabellenkalkulation	
Tabellenkalkulation-Datei: C:\	Bul
Tabelle: Table1	
Daten-Vorschau	
Anzahl gefundener Zeilen: 4	
Gefundene nutzbare Spalten: Da	tetime
Tabellenkalkulation-Datei-Vorsch	au
1: Datetime	
01.04.2021 00:00:00	
15.01.2022 15:04:07	
15.06.2009 13:45:30	

A Here are various data formats that are recognized as dates, among other things.

to see that.

ten Vorschau					
nzahi gehandener nutubarer PCF-Formul		h	derungsdatta	m 65.12.2021	19:1442
f Verschau					
Automatical	Feldt Vorname Feldtyp: TextBox	bearbeiter	Bearbeitet	Scheelbachutz	Ausgewählte Dat
	Schreibschutz	Barbar (2	16		Datatime
- List of called the last	Datenquelle	Barbar 📴			
11	Keine	Barban (2			
• • [T] Aa &•	Datenquelle nutzen: Datetime	beter 2	0	0	
Deispier-	Datum / Uhrzeit - Formatierung: ddlMMJyyyy	Barbar (2	-	-	
Contraction of the local division of the loc	Konstanter Text:	Series II			
				0	
[Company Name]	Schriftart				
Street Address, City, 5		and D		0	
(Email)					
[wenner]					
TEXTFELD-BEISPIEL (
Vorname Pils2021					
			Automatio	uhe Felddater	Zoweisung
Straffe	Ok 🗸				
	h				
					Territoria de la constante

A The use of the data source with the table column "Date-time" and the date formatting field was specified here. This input field allows you to specify a date and/or time as the desired formatting of the data.

For the exact possible values, see a table in section 6.4. Please

Pay attention to the case sensitivity, as e.g. "MM" stands for month and "mm" for minute.



A According to the format "dd.MM.YYYY", the PDF text field contains the date without time in German format.

3.3.9 Check box



A The form field name "checkbox" and the form field type "Check-box" are displayed here.

B Data source: In comparison to text fields (see section 3.3.4), for example, there is no constant value available as text here, but rather a checkbox that functions in the same way. When it comes to the data source, it is important to ensure that values are used in the column that BulkPDF Business converts to a "Yes / No" or "No" answer.

"True / False" value can be converted. (See section 1.4.2 for accepted values)

3.3.10 Radio buttons

BulkPDF Business			– ø ×
BulkPOF Business	Determining Determining PDF Determining Pelotype PDF Determining		
Jan 15 Kr 1927	CONSIGNER LISTERN	Ot v	Interference (E. C. Secondaria)
Rechtlichen 🖶			
Copyright & 2021 Auton Barters	Support 53 Kanfiguration Laden # [] Spectrum 2()		- Zurück Weiter -

A The form field name "radiobutton" and the form field type "Ra-dioButton" are displayed here.

B Data source: As with text fields (see section 3.3.4), a data source, a constant value or no value can be selected. To select which of the radio buttons in a "RadioButton" group is to be selected, a number must be specified. The first box is the value "1", the second the value "2" and so on.

3.3.11 List box

BulkPDF Business		- 0	×
BufkPOF Business	Description PEE formular Optimizer Perspective PEE Description Description Description Description PEE Description Description Feldings: Listics: PEE Description PEE Description Description Feldings: Listics: PE Description PEE De	Austr berougsdater: ELF 2013 0756.34 Deter backette: Backette: Ausgestähle Dates Backette: 2 Backette:	
Rechtliches 🖤			
Casyright & 2027 Julies Barters	Support 23	- Zurititi West	der -

A The form field name "listbox" and the form field type "ListBox" are displayed here.

B Data source: As with text fields (see section 3.3.4), a data source, a constant value or no value can be selected. Another option is "Field data as constant". In this selection, a value can be selected that was intended by the creator of the PDF form for this "ListBox" and is included as a selection for this in the PDF.

The selected value then behaves in the same way as a self-written text in the "Constant text" section. The value is therefore the same for all PDF forms created.

3.3.12 Combo box

BulkPDF Business	C2							- 0 ×
	Satala Dangat	a 🔅 POF Formular 🔄 Data						
	PDF-Datelauseabl							
	PDF-Dates Cristers		OF Example pdf					Autophier.
	Dates Vorschau							
a 1	Anzahl gefundiener nutzbarer PDF-Form	sular Felder: 8	Dokumententyp: Acculture		Entellungsdatum: 07.02.2012 17.47.09		ederungsdatum: 03.07.2015.07.5	634
	PDF-Verschau Automatio	Datenzuweisung 🗟				Wart	Daten bearbeiten Bearbeit	et Ausgewählte Daten
	DOM: NOT BOTTLE	t: combobox		Feldtyp: ComboBox			Beatatan (2)	(P)ucher
		(A)				_	Beatatas 👔 🗌	
		Datenguelle					Beatieter 🔮	
BulkPDF Business		Keine:					Bearbeiter 🖉 🗌	
	TEXTBOX 1. Ruther	O Datenquelle nutzen:					Bearbeiter 🔮 🗌	
		O Konstanter Text:					Beatleter 🖉	
POP		Enddates als Kesstastes Kurben						
	OHEORBOX	(B)						
		Schriftart						
202111-16-1757	RADIO BUTTON							
	COMBOBOK							
	LISTBOX					Ok 🖌		
		P		1111				• Histologen
					Falatyp Salta	Peatiest	Bescheelbung D	sten bearbeiten Daten bearbeiten
	Bulk	DE Rusiness D	NEMO					
Datumentation (2)	DUIKP	OF Business D	EMO					
	(Lean	n more on Bull	kPDF de)					
Rechtlichen 🖤								
	Support ISI		Konfiguration	Later F [2] Spect	div re			- Zurück Weiter -
Copyright & 2027 Julien Barters	Transmission and the second se					BulkPDF Business		Sector Se

A The form field name "combobox" and the form field type "Com-boBox" are displayed here.

B Data source: The data source selection in the "ComboBox" is the same as in the "ListBox", see section 3.3.11.

3.3.13 Barcodes in general

Under the menu item "Additional special fields" there are a number of barcodes that can be used in PDF forms. These include QR, DataMatrix, PDF417, EAN8, EAN13, Codabar, Code11, Code39, Code93, Code128[A,B,C], GS1-128 and UPC. Each of these barcodes has its own interface that is specific to the barcode with its own setting options, but is similar to the QR code interface presented in chapter 3.3.14.

3.3.14 QR code

BulkPDF Business			- 0 ×	
BulkPDF Business	Determine Determine Determine Determine Fill Annual Fill Annua<	10 Höher 102 <th>Annerez</th>	Annerez	
	Lattox		· Hostiger	
Dataser Star B	5.	Policy Solite 1 PolicyRenovie 1 (Kol), Yo	Padian Beckellung Data Seebelin Data Seebelin Q PerCharcole 0.0 Beckellung (industry)	
Bachtlaben 🖤				
Copyright & 2021 Julies Barters	Support 102 Kandigan	alian (Later F) Spectrum (2)	- Zarick Weter -	

A The field type "PdfQRBarcode" is displayed here.

B Options (width, height & position): Here you can set the width and height of the QR code, as well as the page on which the barcode should be displayed and the position on the page. To determine the position, the distance to the 0 point (the top left corner of the PDF page) is specified. The distance to the 0 point is also measured at the top left corner of the QR code. To find the optimal position, you currently just have to play around with the values to get a feel for where, for example, a value of 100 is on the X position in the PDF.

C Options (special setting options): This shows options that are specific to a barcode type, such as the "Quiet Zone", "XDimension" and the "ErrorCorrectionLevel".

D Data source: The data source selection otherwise behaves like that of a "text field", see section 3.3.4. The only difference is in the use of the data, since the data passed to the field is converted to the barcode become.



A The QR code on the PDF form with the values width "100", height "100", Page "1", X-position "100" and Y-position "100".

B Field type: In this case "PdfQRBarcode".

C Page: Number of the PDF form page on which the barcode is to be displayed should.

D Position: Here you can see the position in short form.

E Description: Here you can see a short description of the field.

F At this point the special field can be edited or deleted again. the.

3.3.15 Hyperlink

BulkPDF Business		- a ×
ButkPOF Business Butk POF Business Jack III 16.057	Determine Determine POF Determine Determine profile Por Optimizer X-Position: 0 Position: V-Position: 0	Aunstien Annetien Mentenen Barbete Augunüte Deter Besteinen 2 9 Kurben Besteinen 2 9 Kurben Besteinen 2 0
	BulkPDF Business DEMO (Learn more on BulkPDF de)	Vallan Bashadang Data bashada Data bashada B Ngarini 8.0 (Bashadan 2) (Anton X)
Copyright © 2021 Julien Barters	Support 33 Kantyportion: Later # [2] Spectrum x[2] BulkPOF Busin	es Weter -

A Options: Here you can set the page and position on which the hyperlink should be displayed.

B Text: Here you can set the text for the hyperlink. The data source selection otherwise behaves like that for a "text field", see section 3.3.4.

C URL: Here you can set the URL or web address to which the hyperlink refers.

3.3.16 Image

BulkPDF Business	- σ ×
Bulk PDF Business	
Berteinen B	
Copyright & 2021 Julies Barries	Konfiguration: [Later # []] [spectrum of]] [Ventrum of]] [Ventru

A Options: Here you can set the width and height, as well as the page on which the image should be displayed and the position on the page. To determine the position, the distance to the 0 point (the top left corner of the PDF page) is specified. The distance to the 0 point is also measured at the top left corner of the image. To find the optimal position, you currently just have to play around with the values to get a feel for where, for example, a value of 100 is on the X position in the PDF.

B Data source: Image fields accept file paths to an image as data. Various image file formats, such as PNG, are supported. The data source selection otherwise behaves like that for a "text field", see section 3.3.4. There is also a quick selection button for selecting an image for a constant file path.
	Barbarba - Dalongarba	POF Formular	ania tylene Petgest		E Mail Entroy Figure also					
	PDF-Dateiaumahi									
	PDF Dates . (Proceedings) pdf									
	Para Banka									
	Land Contractor									
	Arcald gelandene natibare PDF Forma						1	Coller ungestant		of the search
	PDF-Verschau	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					-			
	Astornation	Feldtyp: Attachment					arbeiten	Bearbeitet S	cheellochutz	Auspradhh
BulkPDF Business		Datenquelle					then E	16		Voriana
	If if the billion	C Keine					then 🖻	8		Nachrame (Textbox)
		O Datenquelle nutzen:					ten 22	8		Scale
		Konstanter Text: C:\Users'	vallpaper.jpg			*	then BE	8		Haussummar
			Anhannedatai unn das Easte	latta augustilan			an R	10	10	Podletzahl
	Beispiel		Annangsuater von der resu	atte auswariten			100 m	100	80	04
		A)						100	0	Cescreter plane sure
								2	0	Australia Results for
							1.0		0	December Tetrated Co.
	[Company Name]									
	[Phone]									
	[Enal] Distant									
	TEXTFELD-BEISPIEL (
								Automat	tiche Felddah	en-Zusseisung
	Vorname here					Ok 🗸				
					-	al.				
	Straffe me me my	Hausson	mer D		hiller feb				-	I free bootstern
					PetOffaccale 1	(X=450, Y=1	150	NOR	code .	Seatates (2
Datumentation	Protestante State	Out Tests			Attachment 1	(K+0, Y+0)		Attachma	-	Bastatan (2)
Rechtliche Trimetier	resourced and			-						
Compright © 2022	and the second se			- 10						
Table of the Party	popper di		Romagerations Lader	- all open						- Detica

3.3.17 File attachment

A File attachment fields accept file paths as data. The data source selection otherwise behaves like a "text field", see section 3.3.4. In addition, there is a quick selection button for selecting a file for a constant file path.

3.4 Options

In this step, the file name generation is defined by stringing together file name formatting elements, and the output directory for the completed PDF files is defined. Optional options are also available, such as global write protection, file compression, data source line selection, and font and PDF signature settings.

BulkPDF Business					– ø ×
	Startseite – Datenquelle –	PDF-Formular Optionen	- Fertigstellung		
	Dateiname)
	Formatierungs-Element	Daten	Daten-Vorschau	Dateinamen-Formatierungs-Elemente	
	A X Datenquelle	Last name	Mueller	Konstante:	Hinzufügen
	Konstante	.pdf	.pdf		
				Datenquelle:	 Hinzufügen
1				Zeilennummer: Hinzufügen Datum / Uhrzeit: dd.MM.yyyy	Hinzufügen
	A)			<u> </u>	
BulkPDF Business	Feinamen-Vorschau 2: Mustermann odf		Datein	namen-Vorschau 3: Musterfraundf	Liste aller weiteren Dateinamen
buikt bit business	C		, [,		
	usgabeordner: C/Users/				Auswählen.
PDF	D)				
	9				
2021 11 16 1757					
2021.11.16.1757					
	Optionale Optionen	~			
	aler Schreibschutz:	n: In: In: In: In: In: In: In: In: In: I	4):		
	Schriftart			Signatur	
	Standard			Signatur Datei	Augustilan
	Under construction	on		signatur-Late:	Auswanien.
	Standard Unicode-Schriftart			Signatur-Passwort	Prüfen
Deloumentation (7)	Windows-Schriftarten:		- Schriftartgro	DBe:	
	Eigene Schriftart:		Auswählen.		
Rechtliches 🚭	(H)			—(I)————	
	Suggest Fill		Konfiguration:	[70] Snaicham UEP	- Zunick Weiter
Copyright © 2021 Julien Bartels	support tes		Koninguration: Laden Kj		- Zuruck Weiter -

A Filename: This table contains all filename formatting elements which, in descending order, form the generated file name.

B Filename formatting elements: This element group allows the filename Formatting element table on the left side to add more elements gene.

C This preview shows the file names generated based on the formatting elements.

D Output: Here you specify the output order in which at the end of the process all completed documents should be added or saved.

E Global write protection: This option allows you to set all PDF form fields to readonly with a single click so that no further editing can be carried out on these.

F File compression: This option activates the maximum possible compression on the created PDF files. This option to save storage space is This is particularly useful if you plan to send the PDF files by email.

G Line to document: This field allows you to limit the lines of the data source that are to become completed documents. The syntax is either writing the individual data source lines separated by commas (example 3,5,6,8) or selecting a range using the minus sign (example 32-48). Both methods can be combined (examples 4-9,34,42-45,56).

H Font: In this expandable element group, the font for all fields can be specified.

I Signature: In this expandable element group, a digital signature can be specified for the document.

Password protection (view document): In this expandable element group, a password can be selected for each PDF via the data source or a constant password for all documents. The password must be entered when opening the

PDF document to decrypt and view it. can.

Password protection (edit document): This option allows you to set a password that is required to edit the PDF document.

This means that access to editing functions such as changing content, filling out forms or adding annotations is only possible if the correct password is entered.

There are different permission levels that can be set to restrict access to certain editing functions. These include:

- **Print:** You can prevent the document from being printed if or only low quality printing may be allowed.
- Edit content: Editing of the content can be completely prohibited so that neither text nor graphics can be changed.
- **Copying content:** You can prohibit copying text or other content from the PDF.
- Edit comments: The permission to add or edit comments or annotations can also be restricted.
- Fill in form fields: You can specify that form fields are visible but cannot be filled in.
- Accessibility: Copying content for accessibility purposes may be prohibited.

Compile document: It can be prevented that the document

ment is composed from various sources.

• High quality printing: You can prohibit printing the document in high quality.

her quality, so that only printing in low resolution allows is.

However, these permissions are only effective if a password has actually been set. Without a

password, the specified restrictions have

no effect and the document can be accessed by any person without

Restrictions can be edited.

BulkPDF Business	Statissis Datenquella PDF-Formular Optionen Datenquelle Image: Control of the state of the	Fertigstallung Daten-Vorschau Usteinamen-Formatierungs-Elemente Eusteinamen-Formatierungs-Elemente Eusteinamen-Vorschau 3:	Hinzufügen - Hinzufügen I Uhrzeit: ed.MM.yyyy Hinzufügen I te aller weiteren. Dateinamen Auswählen.
	Optionale Optionen Globaler Schreibschutz Datei-Kompression: Zeilen zu Dokument (Beispielt),4-9,3	14):	
	Under construction	Signatur-Datei:	Auswählen.
	Standard Unicode-Schriftart	Signatur-Passwort:	Prüfen
Dokumentation	O Eigene Schriftart:	Aupwählen.	
Rechtliches 🚭			
Copyright © 2021 Julien Bartels	Support 🖾	Konfiguration: Laden 또열 Speichern 뇌열	- Zurück Weiter -

A Here are the options to place a formatting element in the order of Position up or down. There is also a button to remove remote from the element.

B Formatting element: This column indicates the element type, such as e.g. "data source" or "constant".

C Data: Here you can see a representation of the data of the selected formatting element.

D Data Preview: This column shows a data preview depending on the element type. For example, in a data source, the data of the first

row of this data source column. For constants, simply the constant text itself.

E Constant: A constant is a constant text that is the same for every PDF form.

F Data source: The data source is different for each PDF form and depends on the data in the respective row of the data source.

G Line number: This number is the line number of the data source.

H Date / Time: This formatting element allows you to add a date and/or time in the desired format to the file name.

For the exact possible values, see a table in section 6.4.

I File name preview: This preview uses the first three lines to

a data source file name is generated.

J List of all other file names: Using this button, you can preview all the file names that would be

 Formular og former X Consult X Consult 	et Dates Last same pdf	Dates Versitias Muster pdf	Datelaanse Formalierunge Renante Konstante	Hoselagen
• X Interpole • X Interpole	Latinare of	Matter pd	Kanatartar	Houdiger
· ·	ul.	**		
[
(Datesquefe	- Hingdagen
	Zeilennummer	Dateinam	ie .	
	1 Mueller.pdf			
namen Verschau I: Mueller pdf	2 Mustermann.pdf			
same Newbork & Mademan of	3 Musterfrau.pdf			Liste aller variteren Deteinanten
	5 Groß.pdf			
deather Cither	6 蛋糕.pdf			Aunthin.
	7 ropr.pdf			
	8 L [, é, ć .pdf			
nak Tylenan de Skrekelat 📄 🗍 Deni Kom				
Schriftert		Schließen		
				Contraction of the local division of the loc
under constructi	on		Speci cen	Automation,
under constructi	on		Spate cen	Rubber
under constructi	on	Schefturspeller	Spate Reset	[Aundrim.] [Filter]
Under construction	on	- Shiftergelle	Sprate Research	[Australian
	anne Verslan 2 Verslan gif anne Verslan 2 Verslan gif den den Correction and Optimum and Optimum	2 Musterman.pdf 3 Musterman.pdf 4 Cen.pdf 5 Giol.pdf 6 Eff.pdf 7 orsport 7 orsport 8 V AY	A Muterman.pdf 3 Muterman.pdf 3 Gospdf 5 Gospdf 6 If Epdf 7 Toroproff 1 4 2 odf Company 1 4 2 odf	2 Mutemano.pdf 3 Mutemano.pdf 4 Kon.pdf 5 God.pdf 6 Iffle.pdf 7 rors.pdf 1, f, é, é pdf

In this window you can see all the file names that would be generated based on the data source. In the first column (A) you can see the line number of the data source and in the second column (B) you can see the generated file name.

3.5 Completion

On this last page of the application, you can see the settings you have made and a list of all the files that will be generated. You can also generate individual files as a preview.



A Preview: This preview shows the settings made as well as a list of all documents that would now be generated.

B Completion: This table lists the PDF documents created.

C Finish: This button generates the completed PDF form files according to the selected settings. After the PDF files have been processed, the output folder is opened automatically.

	Vorschau				Fertigstellung				
	Ausgabeordn	er: C:\Users\			Zeilennummer		Datei	Fehlerinformationen	Anzeige
	Anzahl gefund	dener Zeilen: 8	Schreibschutz: Datei-Kompression:		1	C:\Users	Mueller.pdf		0
	Anzahl an PDI	F-Formular-Felder: 6	Anzahl an Spezial-Feldern: 0		2	C:\Users	Mustermann.pdf		0
					3	C:\Users	"Musterfrau.pdf		۵
	Zeilennummer	6111 - 1	Datei	PDF-Vorschau	4	C:\Users	,Klein.pdf		۲
	1	Chusers	Mueller.pdf	0+0	5	C:\Users	\Groß.pdf		0
	2	Chusers	Mustermann.pdf	D+0	6	C:\Users	蛋糕.pdf		0
ness	3	C:\Users\	Musterfrau.pdf	0→∞	7	C:\Users	,ropt.pdf		0
	4	C:\Users\	Klein.pdf	0→@	8	C:\Users	ıf, î, ê, ĉ.pdf		0
	5	C:\Users\	Gro8.pdf	©→@					
	6	CAUsers\	蛋糕.pdf	0→@					
	7	C:\Users\	торт.pdf	0→®					
	8	C:\Users\	ř, ľ, ě, č .pdf	□→◎	5				
E									

In the completion table **(A)**, the generated files are displayed with their line number and the file path. The green color and the absence of a Error information indicates that all documents were created successfully. The document can be viewed using the eye button. If If an error message is displayed, clicking on the eye button for the faulty line in the pre-filled support window with a more detailed Error message, as well as the option to contact support directly.

	Auroshaordnar: Cillizer		Zeilennumm	Datei	Fehlerinformationen	
	Assolution and asso 7-llass	Cabalhashata D	1	C:\Users' Mueller.pdf		
	Anashi an DDE Esemular Esideri	Anachi to Saturi Saldana O		C/\Users' Gro8.pdf		
		Pricem an apeziar reading a				
	Zeilennummer	Datei	PDF-Vorschau			
	2 CAUser	Mustermann.ndf				
	3 C\User	Musterfrau.pdf				
Business	4 C\User	Kleinpdf	[]→@			
	5 Cl\User	Großpdf				
	6 CAUser	.蛋糕.pdf	(B)			
	7 C:\User	,торт.pdf	[]→@			
	8 Ci\User	ιř, ľ, ė, č.pdf	[]→@			
1 16 1757						
11.06.1757						

This screenshot shows, for example, that line numbers "1" and "5" have already been generated as preview files using the buttons in **A** and **B**. These two lines were successfully created and entered into the completion table in **C**.



In this screenshot you can see the progress window that appears after clicking the "Finish" button. In **A**, the number of the file that is currently being created is shown at the front and the number of all files behind it. **B** shows the file name of the file that is currently being filled.

3.6 Email draft generator

After the PDF files have been completed, there is the option of adding the next step, the email draft generator. This allows you to create emails with the previously generated PDF files and attach them straight away.

These emails are not sent immediately but are saved as files in the email file format ".EML" or ".MSG". These files can be opened as email drafts using email programs such as Microsoft Outlook or Thunderbird.

and, if necessary, sent immediately. The user interface of the email draft generator is discussed below, the options and explicitly on

RulkPDF Business IDEMO ~ B er Zeilen: å ahl an PDF-Formular-Felder: ୕୶ୠ 104 05 2022 M ୕୶ୣ -----**□**→**Q** [**∂**]→**Q** ୕୶ୣ 04.05.2022 Groß Anna-Lisa or Ø→Q, 04.05.2022 蛋糕_蛋糕.pd 04.05.2022 topt_topt.pdf ୕୶ୣ /04.05.2022 f, ľ, ě, ć הקליטה_pdł ē→Q, Alle D A Speichern 🛛 🖾 ĸ₿ - Zurück

the possibilities for formatting emails.

A In the PDF completion step, the Email Draft Generator button adds open" add the additional step "Email Draft Generator".

B After creating the step, you can always open it again using the button "Open email draft generator" or by clicking on the step in the step overview can be achieved.



A This is where the data assignment to the email draft takes place. In this Graphics only sender, recipient, recipient CC, recipient BCC, subject and the importance are presented.

For recipient, recipient CC and recipient BCC, there is a notation for filling in these fields. An email address is only recognized if

if it is written as follows "support@bulkpdf.de<Support Team>". First

The notation is the email address followed by a "<", then the display name and finally a ">".

If several email addresses are addressed

The addresses must be separated by a "|".

For example, like this:

"support@bulkpdf.de<Support Team>julien.bartels@bulkpdf<Julien Bartels>"

For the subject, there is the special feature that you can use the notation data source column name within a text to refer to columns of the same row from the data source The program replaces these with "{" and "}" enclosed Column names are then replaced by the data, as shown in the example where the text "Current positive developments in {location}" to "Current positive developments in a location}" to "Current positive developments in Berlin" in the preview. (The resolution of bracketed texts to data can be prevented by escaping "{" and "}" with a preceding "\" [Example: \{Test\}]. This is especially necessary if

you want to use HTML with CSS in the email.)

The importance is set to "Normal" by default, but you can change the value, with 0 meaning "Not Important", 1 meaning "Normal" and 2 meaning "Important".

B Here you can see the preview of the data that is visible in the A elements

were set. As was the case for the PDF preview, the first line of the data source is used for the data.

C Here you can see the completion group element of the e-mail drafts. There you can select the data type for the e-mail drafts and use the "Generate e-mails" button to create the files. These are then shown in the table, similar to the PDF files, with a green background if they have been completed successfully and a red background if there is an error.



A Settings for the generated PDF file, file attachments and content in text and/or HTML are made here. For the generated PDF file, you can decide whether you want it attached to the email. For the file attachments, there is again a notation where you have to separate multiple paths with a "|", e.g. "C:\ \report.ppt|D:\\Recipes\\Apple pie.pdf". For the paths for the file attachments and for the content in text and HTML, the notation with "{" and "}" can be used again, as was previously the case for the subject, in order to Insert data using the column names from the respective row (see the difference between the email body content in A and B).

B Here you can see the preview of the data that is visible in the **A** elements were set. As was previously the case for the PDF preview, the first line of the data source is used for the data. In the selected email text content, the column names of the data source enclosed in brackets were replaced with the respective data in the line.

BulkPDF Business [DEMO]			-				ø ×
	Startseite -			E-Mail-Entwurf	fsgenerator		
	E-Mail Datenzuweisung			Vorschau			
	Konstanter Test			E-Mail-Emptanger	r CC:		-
	C Konstanter lext.		× .	E-Mail-Empfänger	BCC:		
	Betreff der E-Mail			Betreff der E-Mail	Derzeitige postive Entwicklungen in Berlin		6
	O Keine			Wishelphale day F	A4.00 T		
	Datenqueile nutzen			wichtigkeit der E-	Man c		
	Konstanter Text:	Derzeitige postive Entwicklungen in (Ort)	÷	PDF-Datei:	\04.05.2022 Mueller_Bernd.pdf		1
BulkPDF Business	Wichtigkeit der E-Mail			E-Mail-Anhänge:	.\ProcessExample.pdf		0
1 million (1997)	O Keine			E-Mail-HTML-Inha	alt: Sehr geehrter Herr Mueller, 		
	Datenquelle nutzen				<pre> die derzeitigen Entwicklungen in Berlin sehe ich als a</pre>	außerst postiv. Anbei die Dokumente diesbezüglich. 	
PDF	Konstanter Text:	2	4 7		 br/> Mit freundlichen Grüßen		
	PDF-Datei			E-Mail-Textinhalt:	Sehr geehrter Herr Mueller,		
		Dis source associated, DDE Data into E Mail biosoficiana of			die derzeitigen Entwicklungen in Berlin sehe ich als äußerst posti	v. Anbei die Dokumente diesbezüglich.	
		Die zuvol generiei te PDP-Datei zur E-main ninzurügen: •		L_	Mit freundlichen Grüßen		-
2022.05.01.0820	E-Mail-Anhänge		(B)			
	O Keine		~	tigstellung			
	 Datenquelle nutzen 	a	*	Zeilennummer	Datei	Information	Anzeigen
	Konstanter Text:	\ProcessExample.pdf	÷	1	\04.05.2022 Mueller_Bernd.EML		9
	E-Mail-HTML-Inhalt			2	\04.05.2022 Mustermann_Max.EMI		G
	Keine			3	\04.05.2022 Musterfrau_Marie.EML		Q
	 Datenquelle nutzen 		· · ·	4	\04.05.2022 Klein_Berthold.EML		Q,
		Sehr geehrter Herr [Nachname (Textbox)], >		5	\04.05.2022 Gro8_Anna-Lisa.EML		9
	Konstanter Text:	 die derzeitigen Entwicklungen in {Ort} sehe ich als äußerst postiv. Anbei die Dokumente diesbezüglich	. 	6	\04.05.2022 蛋糕_蛋糕.EML		G
		 br/> Mit freundlichen Grüßen		7	\04.05.2022 topt_topt.EML		Q
Vollversion jetzt bestellen!	C Maril Teatrahada			8	\04.05.2022 ř, ľ, ě, č _הקליטה_EML		9
	E-mail- lextinnait						(L
Dokumentation	O Datenquelle nutzen	c		E-Mail Dateityp: E	ML		-
Bashali da Ulimania 🖻	Kunduntu Tart		14		E Maile conscione		
Necharche Fanweise ge	A Roussamer lext:			-	e mais generate		
Copyright © 2022	A.						
	The second se	Konfiguration:	aden K 🖾 🛛 🖇	peichern 🖌 🖾		= Zurück	

For HTML content (see A), this is automatically written into the text content (see B) without the HTML tags if no text content is specified.

3.7 Miscellaneous

This is followed by other windows within the application that cannot be assigned to a specific program step.

3.7.1 Support window

BulkPDF Business [DEMO]				- 0 ×
	Sartelle - Dilengelle			
BulkPDF Business	BulkPDF Busines Vielen Dank, dass Sie sich für die In den nächsten Schritten wird Si Klicken Sie nun bitte, um nu dem Ritte nehmen Sie sich die Zeit ei Klicken Sie hier, um die Dokume Für einen schnellen Überblick ük Klicken Sie hier, um die Beispiel- Falls Sie Kritik, Wünsche an die v	Notane yoon hall PDF Business entrobleden haben. Die Software, wer state Programmer Ausgeheiten einer Datematike der Zuseichen einer der Software Versteilten einer Software Software (Software Software) Portuge (Software) Portuge (Software) Por	$\label{eq:second} \begin{tabular}{lllllllllllllllllllllllllllllllllll$	undlich und datenschutzgerecht für jeden Verfügbar macht. eie der Ausgabe Ihrer PDF-Formular-Dateien führen.
	Software Lizenz	Abbrechen	Senden	
	Software-Lizenz-Code	Lizenz aktivieren	Installiert 20211230371	Verfigher: 2021.12.30.0011
	Lizeno Status: X Verfallsdatum		Versionskanat State +	Auf rece Version attuationen 🔯
Defeneration 🛱	Optionen Sanaka Davitski de			
Rechtlichen				
Copyright & 2021 Julien Barless	Support 52	Readigoration: Last	n f B Specters 20	Weber -

A In this area you can write a message for support.

B This area contains the technical error message.

C A support code is displayed here, which was generated based on the software license.

3.7.2 Documentation window

BulkPDF Business [DEMO]		- 0	×
	Sarturia Delangarila C. PDI Formed	ter - Change	
	BulkPDF Business		
	Vielen Dank, dass Sie sich für die Nutzung von Bulk In den nächsten Schritten wird Sie das Programm zu Klicken Sie nun bitte, um zu dem <u>nächsten Schritt zu</u>	PDF Business entschieden haben. Die Software, welche automatisiertes Ausfüllen von PDF-Formularen benutzerheundlich und datenschutzgerecht für jeden Verfügbar mach ar Auswahl einer Datenquelle, der Zuweisung der Daten zu PDF-Formular Felder, der Dateinamenseinstellungen sowie der Ausgabe Ihrer PDF-Formular-Dateien führen. a gelangen, auf die Schaltfläche unten in der rechten Ecke.	6.
	Bitte nehmen Sie sich die Zeit Klicken Sie hier, um die Dokur $A \stackrel{c}{=} \stackrel{c}{\not {}}_{\sim} I$	1 /50 ▶ ▶1 755 • O O ⊡ ⊠ ₽ ℓ • ℓ • Γ • ∓ • 6 • № IT Aa	
BulkPDF Business	Für einen schnellen Überblick üb Klicken Sie hier, um die Beispiel-		
	Falls Sie Kritik, Wünsche an die v	BulkPDF	
		BUSINESS	
		DOWNSDUTITION	
212112.30 811		DOKUMENTATION	
	(B)—		
		In System PDF-Viewer offnen all	
	Software Lines Code	Liere attrieven biskeliket 2011/12/0377 Verfigher 2011/12/0377 - Verlindent falls * Auf rear throws attributions []	
Dakumentatum 🔂	Optimum		
Reddichen 🔮	Spracher, Doutsch ute		
Copyright & 2021 Julier Bartels	Support 02	Reeligention: Later F [] Spectrer x []	eiter -

A At this point, functions such as zoom, page change, etc. are available. tion.

B The documentation itself is displayed here.

C This button can be used to open the documentation in the system's own PDF viewer.

	Starballa - Dateraged	le 🗢 PDF Formular 📼 Optimum 📼 Fartigatellung		
	Verschau		Fertigstellung	
	Augubeardner: Cilliers		Zelennammer Datei	Fehlerinformationen A
		I wanted I wanted	1	
	Anzahl gehandener Zellen: 200	Schreibschutz		
	Angabi an PDF-Formular-Felder: 6	Anzahl an Special-Felders: 0		
	Con I			L
	Zellennummer	Fehlerinformationen		
	1 C1Users1	System.Exception: Die Demo erlaubt nur 23 Zeilen gleichzeitig.		
	2 Cillert	at BulkPDF_Business_Core.DocumentClasses.PDFFiller.FillPDFAndSave(Configuration Action'1 errorOccured Action'1 newProcessedFilesInformation) in	20211011 rdc, Action'3 updatePercentageAction, Func'1 checklfAbortedAction, DEFiller.csline.47	
	1 (1)	at BulkPDF_Business.Classes.PDFFillerGUIManager.<>c_DisplayClass5_0. <startpdfi< td=""><td>iller>b_1() in Classes</td><td></td></startpdfi<>	iller>b_1() in Classes	
OF Business	Costr.	FillerGUIManager.cs:line 56		
	4 Cilibert	(A)		
	5 Cilliand	Log since application start:		
	A Citized	<e> <t> 22.11.2021 11:49:15 </t> <l> Step </l> <mn> Application_Startup </mn> <sf< p=""></sf<></e>	>App.xaml.cs <ln>33</ln>	
<u> </u>	Course	<t><t><t><t><t><t><t><t><t><t><t><t><t></t></t></t></t></t></t></t></t></t></t></t></t></t>	aantwindow.xami.cs <ln>48/</ln> <m></m> nged <sf>StepLandingPage.xaml.cs</sf> <ln>204</ln> <m></m>	
	7 C1Uvers1	<e><t>22.11.2021 11:49:16</t> <l>Step</l> <mn>cbLanguages_SelectionChange</mn></e>	d <sf>StepLandingPage.xaml.cs</sf> <ln>131</ln> <m></m>	
	8 Ciliteral	<e><t>22.11.2021 11:49:16</t><l>Step</l><min>cbTables_SelectionChanged<!--/</td--><td>NIX<ssf>DSExcel.xaml.cs<ln>74</ln></ssf></td></min></e>	NIX <ssf>DSExcel.xaml.cs<ln>74</ln></ssf>	
<u> </u>	 (1) 	<e><t>22.11.2021 11:49:16</t> <l>Step </l></e>	pPDFSelection.xaml.cs <ln>282</ln> <m></m>	
	* COMP	<e><t>22.11.2021 11:49:17</t><l>Step</l><mn>tbOutputDirectoryPath_TextChi</mn></e>	inged <sf>StepProcessing.xaml.cs</sf> <ln>354</ln>	
	10 Cillierd	<e><t>22.11.2021 11:49:17 </t> <l>Step </l> KIN>gContentTitle_Loaded KIN> KIN>there is a standard with the standard standard with the standard stand</e>	SF>MainWindow.xaml.cs <ln>407</ln> <m></m>	
		<e><t>22.11.2021 11:49:17</t><l>Warning</l><mn>Window_Loaded</mn><s< td=""><td>F>MainWindow.xamLcs<ln>498</ln><m>Application closed not</m></td><td></td></s<></e>	F>MainWindow.xamLcs <ln>498</ln> <m>Application closed not</m>	
		normally.	ndow.xaml.cs <ln>144</ln> <m></m>	
		T>22.11.2021 11:49:20 <l>Info </l> <mn>OnClosing </mn> <sf>MainWi</sf>	ndow.xaml.cs <ln>158</ln> <m>CLOSEAPPLICATION</m>	
		Phal Informationen		
		Support-Schlüssel:	DEMO	
		(C)		
		Fehlerbericht nicht versenden	Fehlerbericht senden	
			2	
Millethen 😨				
	[Andrew Control Control of Control	and the second sec	
	Dispersion and	Konfiguration: La	No 11 January 11	- Zursta Fert

3.7.3 Error window (Exception)

A The error message is displayed here.

B The complete log since the program start is embedded here.

C This text box contains the support code that was generated based on the software license.

3.7.4 Font selection & font color selection

When selecting the font, you can also set the font to bold or italic. For example, the Windows fonts for Arial include the normal "arial.ttf", but also **"arialb.tff"** for bold and **"ariali.tff"** for italic.

m Programm enthaltene Schriftarten: Muss kompatibel mit folgenden Zeichen sein:	~
Windows-Schriftarten: Muss kompatibel mit folgenden Zeichen sein:	- Schriftar
Eigene Schriftart:	Auswählen.(E)

A This is the default font.

B Fonts included in BulkPDF-Business, which contain a suitable font for almost all character systems worldwide. The selection can be restricted through the compatibility search. This works by entering, for example, the Chinese characters that the font must support.

C The fonts included in your Windows. Also with the compatibility search, as with the included fonts.

D You can select your own font file here.

E The step size can be set here

F In this element, the font color can be set, as well as reset to the standard black.

3.7.5 Additional functionalities

Vorschau			Fertigstellung			
Ausgabeordner:			Zeilennummer	Datei	Information	Anzeige
Anzahl gefunden Anzahl an PDF-Fo	rr Zeilen: 8 Schreibschutz: Datei-Kompressio rmular-Felder: 10 Anzahl an Spezial-Feldern: 1	n:				
Zeilennummer	Datei	PDF-Vorschau				
1	\04.05.2022 Mueller_Bernd.pdf	ē→Q				
2	\04.05.2022 Mustermann_Max.pdf	¢⇒Q				
3	\04.05.2022 Musterfrau_Marie.pdf	ē→Q				
4	\04.05.2022 Klein_Berthold.pdf	o→Q				
5	\04.05.2022 Gro8_Anna-Lisa.pdf	©→Q,				
6	\04.05.2022 蛋糕_ 弄 糕_pdf	©→Q				
7	\04.05.2022 topt_topt.pdf	ē→Q				
8	\04.05.2022 ř, ľ, ě, č _noťon.pdf	ē→9,				
				Fehleranalysel	enster offnen	
				Fehleranalysel	enster offnen	
				Fehleranalyset	enster offnen	

Merge all documents and open standard PDF viewer This function

tion is intended for mass printing, for example. If all PDF files are to be printed physically right away, they can be used with this function to create a into a single file and then, for example, using Adobe Acrobat can be printed with a single print job. **E-mail draft generator** For information about the e-mail draft generator, see chapter 3.6.

Command line (BulkPDF-Business Console)

In addition to the graphical application, there is also the console application "BulkPDF-Business-Console.exe" in the installation folder. For the command line function

you can easily create a configuration via the GUI, which you can End saves. The configuration can be easily changed later via GUI or using a text editor, since BulkPDF files are text files.

4.1 Parameter list

Below is a list of all available parameters, which can be entered in any order can be used individually or in combination:

parameter	Description A				
	configuration file from the graphical				
	application is always required.				
wait	Keeps the console open to view the				
Wait	output.				
	Causes the program to ignore errors and fill in as much as				
ignoreErrors	is possible under the conditions. Overrides				
	the path to the spreadsheet file. Overrides the path to the PDF source file. Overrides the output directory path. (Attention: There must be no \ at the end, otherwise the " is not recognized and a Errors such as "The syntax for the file name, Directory names or the Volume label is incorrect") Overrides the table name from which the data is to be obtained.				
overwriteSpreadsheetFilePath "C:					
\PATH\new.xlsx"	the spreadsheet file.				
overwritePDFSourceFilePath	Querri des the notifie to the DDE services file				
"C:\PATH\new.pdf"	Overrides the path to the PDF source file.				
	Overrides the output directory path.				
	(Attention: There must be no \ at the end, otherwise				
overwriteOutputFolderPath	the " is not recognized and a				
"C:\PATH"	Errors such as "The syntax for the file name,				
	Directory names or the				
	Volume label is incorrect") Overrides the table				
overwriteSelectedTableName	name from which the data is to be obtained.				
"NewTableName"					
	Tried all form fields without				
	to automatically assign a column from the data				
useAutoMappingForFieldData	source to the assigned data.				
2	Takes a number for the allowed				
	Levenshtein distance between				
	Form field names and column names.				
	After creating the PDFs, executes the email				
emailGeneration	generation according to the information in the				
	configuration file.				



4.2 Output of the console application

In this screenshot you can see the output when no program arguments to call the "BulkPDF-Business-Console.exe" . **A** sees you can see the program version, below you can see help on what the program arguments or parameters for executing a configuration look like. In addition to The path to the BulkPDF configuration file can also be specified with "wait" to keep the console open to view the output. Likewise There are other parameters that were previously listed in the table. a few examples of how to call the console application:

```
BulkPDF-Business-Console.exe "C:\PATH\Config.BulkPDF"
1
      ...
2
     BulkPDF-Business-Console.exe "C:\A\Config.BulkPDF" wait
3
      ...
4
     BulkPDF-Business-Console.exe "C:\B\Config.BulkPDF" ignoreErrors
5
      ...
6
     BulkPDF-Business-Console.exe "C:\Config.BulkPDF" wait ignoreErrors
7
      ...
8
```

90 Manualt Basel Balan Dringging Canadr	-	×
		Â
2021.11.22.1638		
<pre> Check command line arguments Load BulkPDF Business license Load configuration file Start filling process {"File":"C:\\Users\\1.pdf")"Message":null,"RowID":1,"Succeed":true}</pre>		
<pre>{"File":"C:\\Users\\</pre>		
<pre>{"File":"C:\\Users\\Left":"C:\\Users\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Users\\Left":"C:\\Left":"C:\\Left":"C:\Left":</pre>		
{"File":"C:\\Users\\ users\\users\users\\users\\users\\users\\users\\users\\users\users\\users\\users\\users\users\users\users\users\users\\users		
<pre>{"File":"C:\\Users\\</pre>		
{"File B:\\Users\\		~

A Support Code: The line shows the support code if a valid license is loaded and the text "Demo" if there is no valid license. Please add the code in the subject line of support emails.

B File: Path to the completed newly created PDF form.

C Message: Contains an error text in case of errors and normally the value "null".

D RowID: Displays the row number of the data source used for the respective completed PDF form.

E Succeed: Indicates whether the filling was successful. ("true" / "false")

C:\Users\use	—	×
{"File":"C:\\Users\\t ube:"}}###################################		
{"File":"C:\\Users\\t ab:"/loc aset/"#1874.pdf","Message":null,"RowID":874,"Succeed":true}		
{"File":"C:\\Users\\t abl²:"Inverse and the set and the state of th		
{"File":"C:\\Users\\t ube"; file": "B76.pdf","Message":null,"RowID":876,"Succeed":true}		
{"File":"C:\\Users\\t abe"/localesty #// B 77.pdf","Message":null,"RowID":877,"Succeed":true}		
{"File":"C:\\Users\\t abe"; face": ####################################		
{"File":"C:\\Users\\t ube"} for the state of the state of		
{"File":"C:\\Users\\t abe"/ Incomentation 880.pdf","Message":null,"RowID":880,"Succeed":true}		
{"File":"C:\\Users\\t ub:"		
{"File":"C:\\Users\\t ub:"/!buc_ust; "##82.pdf","Message":null,"RowID":882,"Succeed":true}		
{"File":"C:\\Users\\t ab:"/!Massage ":null,"RowID":883,"Succeed":true}		
{"File":"C:\\Users\\t ub:"		
{"File":"C:\\Users\\t ub-1,1aa-1,4aa ###################################		
{"File":"C:\\Users\\t ube: "Land the second se		
{"File":"C:\\Users\\tage":auto: #93.pdf","Message":null,"RowID":893,"Succeed":true}		
{"File":"C:\\Users\\table": # # # # # # # # # # # # # # # # # # #		
{"File":"C:\\Users\\table": True}		
{"File":"C:\\Users\\takes\\Users\\takes\ takes\ takes		
{"File":"C:\\Users\\tage":null,"RowID":897,"Succeed":true}		
{"File":"C:\\Users\\tage:"Succeed":true}		
("File":"C:\\Users\\tage":null,"RowID":899,"Succeed":true}		
Finished		
A ess Any Key to Continue		
		 ~

A shows the message that appears when all PDFs have been successfully filled out and created .

III Microsoft Total Design Principles Seconds	— ·		\times
ileMode mode, FileAccess access, FileShare share, FileOptions options, Int64 preallocationSize)\r\n	at System	.IO.F	ile. ^
Create(String path)\r\n at BulkPDF Business Core.DocumentClasses.PDFTargetAcroforms.Save() in C:\\Us	ers\\tade	l\\Git	chub
\\BulkPDF-Business\\BulkPDF-Business-Core\\DocumentClasses\\PDFTarget\\PDFTargetAcroforms.cs:line 80\r	\n at B	ulkPDF	Bu
siness Core.DocumentClasses.PDFFiller.FillPDFAndSave(Configuration20211011 rdc, Action`3 updatePercent	ageAction	, Fund	1
checkIFAbortedAction, Action`1 errorOccured, Action`1 newProcessedFilesInformation) in C:\\Users\\tade	1\\Github	\\Bulk	KPDF
-Business\\BulkPDF-Business-Core\\DocumentClasses\\PDFFiller.cs:line 109", "RowID":877, "Succeed":false}			
{"File":"C:\\Users\` raisI' face last ("ar 878.pdf","Message":null,"RowID":878,"Succeed":true}			
{"File":"C:\\Users\` rate I'dere last 'o"""879.pdf","Message":null,"RowID":879,"Succeed":true}			
{"File":"C:\\Users\` rais in Base lasis "###################################			
{"File":"C:\\Users\` main" in Base Instit " #81.pdf", "Message":null, "RowID":881, "Succeed":true}			
{"File":"C:\\Users\` ruteI": "Succ eed":true}			
{"File":"C:\\Users\` nate I' fine in 1 883.pdf", "Message":null, "RowID":883,"Succeed":true}			
{"File":"C:\\Users\` rues in Section 1 884.pdf","Message":null,"RowID":884,"Succeed":true}			
{"File":"C:\\Users\\ "unter information 1885.pdf", "Message":null,"RowID":885,"Succeed":true}			
{"File":"C:\\Users\\ "sets": "886.pdf","Message":null,"RowID":886,"Succeed":true}			
{"File":"C:\\Users\\ "sets in the file and t			
{"File":"C:\\Users\\ "unter file": "C:\\Users\\ "unter file": "C:\\Users\\ "unter file": "C:\\Users\\ "unter file ":"C:\\Users\\ " "C:\\Users\\"unter file":"C:\\Users\\"unter file":"C:\\Users\"unter file":"C:\\users\"unter file":"C:\\users\"unter file":"C:\\users\"unter file":"C:\\users\"unter file":"C:\\users\"unter file":"C:\users\"unter file":"C:\users\"unter file":"C:\\users\"unter file":"C:\users\"u			
{"File":"C:\\Users\' Hann Three land "889.pdf", "Message":null, "RowID":889, "Succeed":true}			
{"File":"C:\\Users\\ "unter "Breelingt": "890.pdf","Message":null,"RowID":890,"Succeed":true}			
{"File":"C:\\Users\\ "unter "Grand Constitution" 891.pdf", "Message":null, "RowID":891, "Succeed":true}			
{"File":"C:\\Users\' uuta 1 Guellant 10, 1892.pdf", "Message":null, "RowID":892, "Succeed":true}			
{"File":"C:\\Users\\ "ustral Tobar Last 1 , "Resage":null, "RowID":893, "Succeed":true}			
{"File":"C:\\Users\\ rue} "."C:\\Users\\ rue "."".""."".""."".""."".""."".""."".""."			
{"File":"C:\\Users\\"Here': The sage "in the			
{"File":"C:\\Users\\= The sage and the sage and the sage "inul, "RowID": 896, "Succeed": true}			
{"File":"C:\\Users\\Therefore the second s			
{"File":"C:\\Users\\ Title": "C:\\Users\\ Title ":"C:\\Users\\ Title ":"C:\Users\\ Title ":"C:\\Users\\ Title ":"C:\Users\\ Title ":"C:\\Users\\ Title ":"C:\Users\\ Title ":"C:\Use			
File : C: (Users) - C: (Users)			
A Section Finished with Errors			
A ress any key to continue			× ×

At **A** you can see the message ("Finished with error"), which appears when not all PDFs could be created successfully. The error must then be looked for in the lines above. It should also be mentioned that the exit code of the console application is 0 if there were no errors and 1 if at least one error occurred and therefore at least 1 PDF file could not be created.

So you can, for example, check in a batch file at %errorlevel% to see if there were any

errors.		
III Mineral-Visal Isale Ostapping Result		×
<pre>{"File":"C:\\Users\\The III and IIII 871.pdf", "Message":null, "RowID":871, "Succeed":true} {"File":"C:\\Users\\The IIII 872.pdf", "Message":null, "RowID":872, "Succeed":true} {"File":"C:\\Users\\The IIII 873.pdf", "Message":null, "RowID":873, "Succeed":true} {"File":"C:\\Users\\The IIII 874.pdf", "Message":null, "RowID":874, "Succeed":true} {"File":"C:\\Users\\The IIII 875.pdf", "Message":null, "RowID":875, "Succeed":true} {"File":"C:\\Users\\The IIII 875.pdf", "Message":null, "RowID":875, "Succeed":true}</pre>		^
<pre>wpportCode: A her in Zeile 877. Datei konnte nicht erstellt werden. [The process cannot access the file 'The state of the state of</pre>	the file 2.SafeHar options) eShare sh ing path em.IO.Fi del\\Gith BulkPDF on, Func ub\\Bulk	e hdl. har le. hub _Bu Y 1 PDF
Business\\BulkPDF-Business-Core\\DocumentClasses\\PDFFiller.cs:line 109", "RowID":877, "Succeed":false} File":"C:\\Users\\tadel\\ Barland in 878.pdf ", "Message":null, "RowID":878, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 879.pdf ", "Message":null, "RowID":878, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 880.pdf ", "Message":null, "RowID":880, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 882.pdf ", "Message":null, "RowID":881, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 882.pdf ", "Message":null, "RowID":882, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 882.pdf ", "Message":null, "RowID":883, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 883.pdf ", "Message":null, "RowID":883, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 884.pdf ", "Message":null, "RowID":884, "Succeed":true} {"File":"C:\\Users\\tadel\\ Barland in 885.pdf ", "Message":null, "RowID":885, "Succeed":true}		

A Support Code: Please add the code in the subject line of support emails.

B Here you can see the error text in "Message", which can be sent as a copy to support if necessary.



Configuration files

This file is encoded in JSON format[5], reasonably readable for humans and can be edited manually with any text editor if necessary. To analyse the functionality and possibilities, the example configuration (see Section 2.1).

5.1 Relative paths

Relative paths[6] can be used in the configuration file . Environment variables, such as %USERPROFILE%, are resolved by BulkPDF Business. Following from the example configuration (see section 2.1), relative paths that are used for the Data source, PDF form and output directory can be used:

```
      1
      ...

      2
      "DataSource": {

      3
      "$id": "40",

      4
      "$type": ... ,

      5
      "filePath": ".\\ProcessExample.xlsx"

      6
      },

      7
      ...
```

1		
2	"PDFSource": {	
3	"\$id": "48",	
4	"\$type": ,	
5	"Author": "",	
6	"CreationDate": "2021-12-01T14:14:19",	
7	"Creator": ,	
8	"IsXfa": false,	
9	"Keywords": "",	
10	"ModificationDate": "2021-12-03T19:14:42",	
11	"Producer": ,	
12	"Subject": "",	
13	"Title": "",	
14	"SourceFilePath": ".\\ProcessExample.pdf"	
15	},	
16		



5.2 Example excerpts from a configuration file

Below are some example extracts, excluding "DataSource" and "PDFSource", because these have already been dealt with.

5.2.1 NormalFields

1	
2	"NormalFields": [
3	{
4	"\$id": "2",
5	"\$type": ,
6	"FieldType": "TextBox",
7	"CurrentValue": "",
8	"FieldSettableOptions": {
9	"\$id": "3",
10	"\$type": ,
11	"ValueSource": {
12	"\$id": "4",
13	"\$type": ,
14	"Constant": "",
15	"DataSource": "First Name"
16	},
17	"WriteProtection": false
18	},
19	"HasBeenEdited": true,
20	"Name first Name",
21	"Preview": "First name"
22	},
23	

5.2.2 SpecialFields



5.2.3 Global options



5.2.4 FilenameGenerator

1	
2	"FilenameGenerator": {
3	"\$id": "41",
4	"\$type": ,
5	"FilenameParts": [
6	{
7	"\$id": "42",
8	"\$type": ,
9	"Preview": "17.12.2021",
10	"Type": "Date / Time",
11	"Data": "dd.MM.yyyy",
12	"Identifier": 8
13	},
14	

Machine Translated by Google

5.2.5 Email

1	"letters": {
2	"\$id": "52"
4	"Stype":
5	"FileType": 0
6	"Sender": {
7	"\$type":
8	"Constant": "julien.bartels@bulkpdf.de <julien bartels="">"</julien>
9	},
10	"Subject": {
11	"\$type":,
12	"Constant": "Current positive developments in {location}"
13	},
14	"Draft": true,
15	"RecipientsTo": {
16	"\$type":,
17	"Constant": "support@bulkpdf.de <support-team>"</support-team>
18	},
19	"RecipientsCC": {
20	"\$type":
21	},
22	"RecipientsBCC": {
23	"\$type":
24	},
25	"BodyText": {
26	"\$type":,
27	"Constant": "Dear Mr. {Last Name (Textbox)},\r\n\r\n
28	I see the current developments in {place} as extremely
29	With kind regards"
30	
32	BodyHTMI ": {
33	"Stype"
34	}.
35	"LetterImportanceValueSource": {
36	"\$type":,
37	"Constant": "2"
38	},
39	"AddAttachmentsGeneratedPDFFilePath": true,
40	"AttachmentsAdditionalsFilePaths": {
41	"\$type":,
42	"Constant": ".\\ProcessExample.pdf"
43	}
44	}
45	



Tips and tricks

A small selection of tips and tricks that will help you create PDF documents can help.

6.1 Create or repair PDF forms with PDFescape

To create and repair public PDF forms, we recommend https://www.pdfescape.com/. For some PDF forms, it may be helpful to upload the PDF once and download it again straight away so that the PDF is again compatible with the most common PDF readers.[4]

6.2 Creating PDF forms with LibreOffice / OpenOffice Writer

LibreOffice is a collection of office software. The software is completely free, compatible with Microsoft Office and is available at http://www.libreoffice.org/ available for download. The software offers the possibility of creating PDF forms.[3] How this works in practice is shown below as an example.



6.2.1 Text field example (text box)

Vorname		Nachname:
Straße:	Ausschneiden Kogieren Einfügen	Strg+X Strg+C Strg+V mmer:
Postleitzahl	Position und Gr Steuerelement-E Formulareigense Ersetzen <u>d</u> urch	öße Eigenschaften ghaften
OPTIONSFELD-BEISPIEL (RAI	🔏 Veran <u>k</u> erung 🔟 Umlau <u>f</u>	:
Geschlecht: Männlic	Ausrichtung	• Divers
MARKIERFELD (CHECKBOX):	Beschreibung	

			Eigenschaften: Textfeld			
TEXTFELD-BEISPIEL (TEXTBOX	.):		Allgemein Daten Ereignisse	e		
Vorname		Na	Name	Vorname		ļ
4 0			<u>B</u> eschriftungsfeld			ĺ
Straße:		Ha	Max. Textlänge	0	÷	l
			<u>A</u> ktiviert	Ja	~	I
Postleitzahl		Or	Sichtbar	Ja	~	I
			Nur lesen	Nein	~	ľ
OPTIONSFELD-BEISPIEL (RAD	OBOX):		Druckbar	Ja	~	I
			Tabulator	Ja	~	Ľ

6.2.2 Radio button example

	v			1			
IEL (TEXTBOX):	x	Ausschneiden	Strg+X				
	0	Kogieren	Strg+C				
	G	Einfügen	Strg+V				
		Position und Größe.					
	00	Steuerelement-Eiger	nschaften				
	88	Formulareigenschaf	ten				
		Ersetzen durch	,	-			
	ę	Verankerung					
	-	Umlauf					
		Ausrichtung					
		Ausrichtung					
	2	Angranung	,				
ISPIEL (RADIOBOX):	Beschreibung						
		Name					
Männlich	3	RIDHCH	DIVERS	Ор			
	-			_			
Postleitzahl		Ort	Eigenschaft	ten: Optionsfeld			×
			Allgemein	Daten Ereigniss	e Optionefield 1		^
OPTIONSFELD-BEISPIEL (RADIOB	OX):		Titel		Optionsfeld		
Geschlecht: Männlich		Weiblich • O Diver	s ^O Op		Optionstela		
MARKIEREELD (CHECKBOX)-			Beschrift	ungsfeld	Geschlecht		1
MANNENI ELD (GILGNDOA).				la			
Umweltfreundliche Verpackung:					la la		
LISTENFELD (LISTBOX):			Deuskhar			-	
Zugehörige Branche:			Tabulator		Ja		
					-		~
KOMBINATIONSFELD (COMBOBO	X):						

Postleitzahl	Ort:	Eigenschaften: Optionsfeld	×
		Allgemein Daten Ereignisse	
OPTIONSFELD-BEISPIEL (RADIOBUS: Geschlecht: Männlich ℃o; Weiblich 🔐 Divers op MARKIERFELD (CHEUKBOX): Umweltfreundlive Verpackung: LISTE NFELD (LISTEX):		Name Optionsfeld 2 Titel Optionsfeld	•
		Beschriftungsfeld Geschlecht A	
			~
Zugehörige Branche:		Tabulat <u>o</u> rJa	~ ,
KOMBINATIONSFELD (COMBOBOX):			_
OPTIONSFELD-BEISPIEL (RADIOBOX):			
Geschlecht: Männlich C Weiblic	h Divers C	Eigenschaften: Optionsfeld Allgemein Daten Ereignisse	
MARKIERFELD (CHECKBOX):		Name rbWeiblich	
Umweltfreundliche Verpackung:		Ţītel	•

6.2.3 Checkbox

OPTIONSFELD-BEISPIEL (RADIOBOX):	Eigenschaften: Markierfeld	×
Carablashi Mizzkish C Mikikish C	Allgemein Daten Ereignisse	
sescniech: Mainlich (* weiblich (*	<u>N</u> ame	^
MARKIERFELD (CHECKBOX):	Titel	
Jmweltfreundliche Verpackung:	Beschriftungsfeld	H
	Aktiviert Ja	
ISTENFELD (LISTBOX):	Şichtbar Ja 🗸	
ugehörige Branche:	Druckbar Ja	
OMBINATIONSFELD (COMB OB OX):	Tabulat <u>o</u> r Ja	
perzeitive Tätivkeit.	Aktivierungsreihenfolge 0	*

6.2.4 List box

Straße:	Hausnummer:
Postleitzahl	X Ausschneiden Strg+X Im Kogieren Strg+C Im Einfügen Strg+V
OPTIONSFELD-BEISPIEL (RADIOBOX):	Steuerelement-Eigenschaften
Geschlecht: Männlich C Weibli	Ersetzen durch
MARKIERFELD (CHECKBOX):	Lengtherung +
Umweltfreundliche Verpackung: 🗖	Ausrichtung
ISTENFELD (LISTBOX):	Beschreibung
tugehörige Branche	
OPTIONSFELD-BEISPIEL (RADIOBOX): Geschlecht: Männlich C Weiblich C Divers	feld X
MARKIERFELD (CHECKBOX): Eingabe erforderlic	
Umweltfreundliche Verpackung:	s Werteliste
LISTENFELD (LISTBOX): Zugehörige Branche KOMBINATIONSEFLD (COMBOROX):	1 Industrial Cake-Factory
Derzeitige Tätigkeit:	Ωĸ
6.2.5 Combo box

sescureen.	Mannuch	weiblich	Divers C	Formulareigenschaften		×
MARKIERFELD (CHECKBOX):			Allgemein Daten Ereignisse		
mweltfreund	iche Verpackung:			Datenquelle	Bibliography	~ [] ^
				Art des Inhaltes	Tabelle	~
STENFELD (LIS	TBOX):			Inhait	biblio	¥
ıgehörige Bra	nche:	•		SQL-Befehl analysieren	Ja	~
MRINATIONS				<u>F</u> ilter		
Moneanonas		-	-	S <u>o</u> rtierung		
erzeitige Tätiș	gkeit:		- <u>+</u>	Daten hinzufügen	Ja	~
	-	-	2	Daten ändern	Ja	~

6.3 Create signature

You can create a signature in Acrobat Reader. For information, see https:// helpx.adobe.com/sign/using/create-electronic-signature.html/ . After you create a signature using Acrobat, you can find this signature as a .pfx file under the path %appdata%\Adobe\Acrobat\DC\Security . This .pfx file can then be used in BulkPDF. (Note the password entry.)

6.4 Date formatting

A formatting string is used to format the date. This string consists of a few characters that define the format. For example, the format **dd.mm.yyyy** is a valid date format that outputs the day, month and year (example: 11/25/2021).

Format specifier	Description	Examples
	The day of the month, from 1 to 31.	2009-06-01T13:45:30 ->1
"d"		
	More information: The "d" Custom Format Specifier.	2009-06-15T13:45:30 ->15
	The day of the month, from 01 to 31.	2009-06-01T13:45:30 ->01
"dd"		
	More information: The "dd" Custom Format Specifier.	2009-06-15T13:45:30 ->15
		2009-06-15T13:45:30 ->Mon (en-US)
	The abbreviated name of the day of the week.	
"ddd"		2009-06-15T13:45:30 -> (ru-RU)
	More information: The "ddd" Custom Format Specifier.	
		2009-06-15T13:45:30 ->lun. (fr-FR)
		2009-06-15T13:45:30 ->Monday (en-US)
	The full name of the day of the week.	
"dddd"		2009-06-15T13:45:30 -> (ru-RU)
	More information: The "dddd" Custom Format Specifier.	
~		2009-06-15T13:45:30 ->lundi (fr-FR)
	The tenths of a second in a date and time value.	2009-06-15T13:45:30.6170000 ->6
"f"		
	More information: The "f" Custom Format Specifier.	2009-06-15T13:45:30.05 ->0

See the following table from the Microsoft .NET documentation[2]:

"##"	The hundredths of a second in a date and time value.	2009-06-15T13:45:30.6170000 ->61
11	More information: The "ff" Custom Format Specifier.	2009-06-15T13:45:30.0050000 ->00
	The milliseconds in a date and time value.	6/15/2009 13:45:30.617 ->617
"ffff"		
8	More information: The "fff" Custom Format Specifier.	6/15/2009 13:45:30.0005 ->000
"ffff"	The ten mousandins of a second in a date and time value.	2009-06-15115.45.50.6175000 ->6175
	More information: The "ffff" Custom Format Specifier.	2009-06-15T13:45:30.0000500 ->0000
	The hundred thousandths of a second in a date and time value.	2009-06-15T13:45:30.6175400 ->61754
"ffffff"		
0	More information: The "fffff" Custom Format Specifier.	6/15/2009 13:45:30.000005 ->00000 2000 06 15T12:45:20 6175420 >617542
"ffffff"	The minioriths of a second in a date and time value.	2009-06-15115.45.50.6175420 ->617542
	More information: The "ffffff" Custom Format Specifier.	2009-06-15T13:45:30.0000005 ->000000
	The ten millionths of a second in a date and time value.	2009-06-15T13:45:30.6175425 ->6175425
"ffffffff"		
	More information: The "ffffffff" Custom Format Specifier.	2009-06-15113:45:30.0001150
"F"	If non-zero, the tenths of a second in a date and time value.	2003-00-15115.45.30.6170000 ->6
	More information: The "F" Custom Format Specifier.	2009-06-15T13:45:30.0500000 ->(no output)
	If non-zero, the hundredths of a second in a date and time value.	2009-06-15T13:45:30.6170000 ->61
"FF"		
51	More information: The "FF" Custom Format Specifier.	2009-06-15T13:45:30.0050000 ->(no output)
"FFF"	ir non-zero, the milliseconds in a date and time value.	2005-00-13113.43.30.0170000 -2017
	More information: The "FFF" Custom Format Specifier.	2009-06-15T13:45:30.0005000 ->(no output)
	If non-zero, the ten thousandths of a second in a date and time value.	2009-06-15T13:45:30.5275000 ->5275
"FFFF"		
	More information: The "FFFF" Custom Format Specifier.	2009-06-15T13:45:30.0000500 ->(no output) 2009-06-15T13:45:30.6175400 ->61754
"FFFF"	in hon-zero, the hundred thousandths of a second in a date and time value.	2003 00 10110.40.00.0110400 201104
	More information: The "FFFF" Custom Format Specifier.	2009-06-15T13:45:30.0000050 ->(no output)
	If non-zero, the millionths of a second in a date and time value.	2009-06-15T13:45:30.6175420 ->617542
"FFFFFF"		
4.3	More information: The "FFFFF" Custom Format Specifier.	2009-06-15T13:45:30.0000005 ->(no output) 2009-06-15T13:45:30.6175425 ->6175425
"FFFFFF"		
	More information: The "FFFFFF" Custom Format Specifier.	2009-06-15T13:45:30.0001150 ->000115
	The period or era.	
"g", "gg"		2009-06-15T13:45:30.6170000 ->AD
5 <u>5</u>	More information: The "g" or "gg" Custom Format Specifier.	2009-06-15T01:45:30 ->1
"h"		
	More information: The "h" Custom Format Specifier.	2009-06-15T13:45:30 ->1
"bbb"	The hour, using a 12-hour clock from 01 to 12.	2009-06-15T01:45:30 ->01
	More information: The "hh" Custom Format Specifier	2009-06-15T13:45:30 ->01
9 <u>5</u>	The hour, using a 24-hour clock from 0 to 23.	2009-06-15T01:45:30 ->1
-н-		
	More information: The "H" Custom Format Specifier.	2009-06-15T13:45:30 ->13
"HH"	The hour, using a 24-hour clock from 00 to 23.	2009-06-15101:45:30 ->01
	More information: The "HH" Custom Format Specifier.	2009-06-15T13:45:30 ->13
0		With DateTime values:
		2009-06-15T13:45:30, Child Unspecified ->
		2009-06-15T13:45:30 Kind Lite ->7
	Time zone information.	
"К"		2009-06-15T13:45:30, Kind Local ->-07:00 (depends on local computer settings)
	More information: The "K" Custom Format Specifier.	
		With Date LineOffset Values:
		2009-06-15T01:45:30-07:00 ->-07:00
		2009-06-15T08:45:30+00:00 ->+00:00
"m"	The minute, from 0 to 59.	2009-06-15T01:09:30 ->9
"m"	More information: The "m" Oustom Format Specifier	2009-06-15T13:29:30 ->29
	The minute, from 00 to 59.	2009-06-15T01:09:30 ->09
"mm"		
	More information: The "mm" Custom Format Specifier.	2009-06-15T01:45:30 ->45
-14-	The month, from 1 through 12.	2000 06 15712-15-20 ~6
IVI	More information: The "M" Custom Format Specifier	2009-00-10110.40.30 ->0
<u></u>	The month, from 01 through 12.	
"MM"		2009-06-15T13:45:30 ->06
	More information: The "MM" Custom Format Specifier	

ý.	The abbreviated name of the month.	2009-06-15T13:45:30 ->Jun (en-US)
"MMMM"		2009-06-15T13:45:30 ->juin (fr-FR)
	More information: The "MMM" Custom Format Specifier.	2009-06-15T13:45:30 ->Jun (zu-ZA)
19		2009-06-15T13:45:30 ->June (en-US)
"MMMM"	The full name of the month.	2009-06-15T13:45:30 ->June (da-DK)
	More information: The "MMMM" Custom Format Specifier.	2009-06-15T13:45:30 ->uJune (zu-ZA)
	The second, from 0 through 59.	
"s"	More information: The "s" Custom Format Specifier.	2009-06-15T13:45:09 ->9
	The second, from 00 through 59.	
"\$\$"	More information: The *ss* Custom Format Specifier.	2009-06-151113:45:09 ->09
	The first character of the AM/PM designator	2009-06-15T13:45:30 ->P (en-US)
"t"	The list character of the Aliver in designator.	2009-06-15T13:45:30 -> (yes-JP)
	More information: The "t" Custom Format Specifier.	2009-06-15T13:45:30 ->(fr-FR)
		2009-06-15T13:45:30 ->PM (en-US)
"tt"	The AM/PM designator.	2009-06-15T13:45:30 -> (yes-JP)
	More information: The "tt" Custom Format Specifier.	2009-06-15T13:45:30 ->(fr-FR)
<u>></u>		0001-01-01T00:00:00 ->1
		0900-01-01T00:00:00 ->0
	The year, from 0 to 99.	
"у"	More information: The "y" Custom Format Specifier.	1900-01-01T00:00:00 ->0
		2009-06-15T13:45:30 ->9
		2019-06-15T13:45:30 ->19
		0001-01-01T00:00:00 ->01
	The year, from 00 to 99.	0900-01-01T00:00:00 ->00
"ууу"	More information: The "yy" Custom Format Specifier.	1900-01-01T00:00:00 ->00
		2019-06-15T13:45:30 -> 19
1		0001-01-01T00:00:00 ->001
	The year, with a minimum of three digits.	0900-01-01T00:00:00 ->900
"ууу"		4000.04.04700:00.00 - 4000
	More information: The 'yyy' Custom Format Specifier.	1900-01-01100.00.00 ->1900
		2009-06-15T13:45:30 ->2009
">>>>>>	The year as a four-digit number.	0900-01-01T00:00:00 ->0900
,,,,,	More information: The "yyyy" Custom Format Specifier.	1900-01-01T00:00:00 ->1900
		2009-06-15T13:45:30 ->2009
	The year as a five-digit number.	0001-01-01T00:00:00 ->00001
ууууу	More information: The "yyyyy" Custom Format Specifier.	2009-06-15T13:45:30 ->02009
"7"	Hours offset from UTC, with no leading zeros.	2009-06-15T13:45:30-07:00 ->-7
-	More information: The "z" Custom Format Specifier.	
"77"	Hours offset from UTC, with a leading zero for a single-digit value.	2009-06-15T13:45:30-07:00 ->-07
<u> </u>	More information: The "zz" Custom Format Specifier.	
"zzz"	Hours and minutes offset from UTC.	2009-06-15T13:45:30-07:00 ->-07:00
	More information: The "zzz" Custom Format Specifier.	
	The time separator.	2009-06-15T13:45:30 ->: (en-US)
"."		2009-06-15T13:45:30 ->. (it-IT)
	More information: The *: Custom Format Specifier.	2009-06-15T13:45:30 ->: (yes-JP)
		2009-06-15T13:45:30 ->/ (en-US)
"/"	I he date separator.	2009-06-15T13:45:30 ->- (ar-DZ)
	More Information: The */" Custom Format Specifier.	2000.06.15T12.45.30 ~ (#LTP)
		2003-00-10113-40.00 P2. (u*11)

1

1

Г			
	"strings"	String delimiter.	2009-06-15T13:45:30 ("arr:" h:mt) ->arr: 1:45 P
	5		
	'string'	More information: Character literals.	2009-06-15T13:45:30 ('arr:' h:mt) ->arr: 1:45 P
h			
		Defines the following character as a custom format specifier.	
	%		2009-06-15T13:45:30 (%h) ->1
ļ		More information: Using Single Custom Format Specifiers.	
	· · · · · · · · · · · · · · · · · · ·	The escape character.	
	1		
	`		2009-06-15113:45:30 (n \n) ->1 n
		More information: Character literals and Using the Escape Character.	
ſ		The character is copied to the result string unchanged.	
	Any other character		2009-06-15101:45:30 (arr hh:mm t) ->arr 01:45 A
		More information: Character literals.	
- L			1

6.5 Connection string

Examples of valid connection strings for a connection with user name and password would be these

Server=myServer;Database=myDataBase;User Id=myUser;Password=myPassword;

and for a "Trusted Connection" via Active Directory login:

Server=myServer;Database=myDataBase;Trusted_Connection=True;

For further examples please have a look at the following website https:// www.connectionstrings.com/sql-server/.

6.6 Simultaneous printing of documents

If all the documents created are to be printed physically immediately, the button "Merge all documents and open standard PDF viewer (e.g. for mass printing)" can be used on the last program page. After further confirmation, this recreates all the documents and combines them into a single PDF document. The created document is then opened in the standard PDF viewer and can be printed immediately from there.

There you can also use the print options to set the pages that should not be printed.

Frequently Asked Questions (FAQs)

7.1 What are the system requirements for BulkPDF Business?

Compatible with Windows 7, Windows 8, Windows 10 and Windows 11. Approximately 500 MB of storage space is required for installation. The program requires a one-time Internet connection for license activation.

7.2 BulkPDF cannot be installed. [Windows SmartScreen]

If a double-click on the downloaded installation file displays a Messages such as "Windows has protected your computer" or "Windows Smartscreen prevented an unknown app from starting" must be Installation clicked on the text "More information" within the window and then click "Run anyway". Such a message is displayed by Microsoft when an executable program has been created relatively recently and is not yet known to Microsoft. Microsoft will automatically upload the exe file in the background, analyze it and only This message will no longer be displayed to others if the program version has already been used on many computers for some time. Therefore, the Unfortunately, this problem occurs again with every new version of the program.

7.3 If my files or data are uploaded or left

my computer in any way?

No! Your data remains on your computer alone. BulkPDF does not download any of your Data is uploaded to a server. The program only establishes a connection to the BulkPDF servers in a few cases. These cases include the one-time license activation, the search for program updates and the check whether a license is currently being used on several computers at the same time.

7.4 Is the program available for Mac OS?

No, but with "Parallels Desktop for Mac" you can run the program on your Mac. (See https://www.parallels.com/)

7.5 How many users may use a license simultaneously? **be detected?**

This depends on the offer selected. Single licenses may only be used by one user at a time, whereas a terminal server license, for example, may be used by several users at the same time, depending on the offer.

may be.

7.6 On how many computers can a license be used?

A license can be installed on multiple computers, but can only be used by one of these computers at a time. When the application is started and during use, it is checked whether the license is already being used on more than one computer. If this is the case, the program is thrown back into demo mode until the computer is again the only one using the software license. Until then, no progress is lost and the program does not close automatically or anything like that. The only difference is that a watermark appears on the PDF documents created again and the number of PDF documents that can be created simultaneously is limited again.

7.7 I get the following message "The license key is currently already used by multiple devices". What does that mean?

This means that the license has currently been used on multiple computers in the last 15 minutes and may not be used on any other computer at the same time. If you have used the license on multiple computers in the last 15

minutes at the same time and the program has now been closed again, you will be able to run BulkPDF Business again on the current computer after 15 minutes at the latest. If this is not the case and you do not know of any other computer on which the program is currently running, license code theft cannot be ruled out. In this case, please contact support so that we can find a solution together so that you can work again.

7.8 In the PDF preview, the form fields are not displayed at the

correct position or shifted

In this case, please use the "Open in system PDF viewer" button and check whether the document is displayed correctly there. In rare cases, the preview may process the PDF files differently than, for example, Acrobat Reader or your web browser.

7.9 No PDF form fields are supported in BulkPDF Business

displayed

BulkPDF requires PDF form elements in the PDF document. If, for example, a PDF document is printed and then scanned again, it is no longer a PDF form document. The PDF document

is then simply a multi-page image, without the technical background data that PDF form documents contain. Without this background data, BulkPDF Business can no longer fill in the fields.

Solution 1 You can overlay the image with a BulkPDF Business special field. For example, you can place a text field, a check box or a radio button as a special field over the image and then fill the special field with your data source. You can find more information about this in section 1.5.4.

Solution 2 You can use suitable software to create new PDF form elements and place them over the fields shown in the image. BulkPDF Business will then find these and allow the fields to be filled in automatically.

7.10 Unicode characters in Acrobat Reader are not displayed correctly (Chinese, Cyrillic, Arabic, etc)

To correctly display Unicode characters in Acrobat Reader, an additional font package is required.

For 32 bit installations, the font package is available at https://helpx.adobe.com/acrobat/ kb/windows-font-packs-32-bit-reader.html to find. For 64-bit installations, the font package is available at https://helpx.adobe.com/ acrobat/kb/font-pack-spelling-dictionary-64-bit-windows.html ready for downloading.

7.11 Windows always opens the wrong PDF viewer

or the web browser (Chrome, Edge) to view PDF files

For example, if the Edge browser or another unwanted browser is always program opens when you double-click on a PDF file, the default PDF viewer is set incorrectly. This can be changed by clicking on a PDF file and click on "Properties" in the menu that appears. In the window that then opens, under the item "Open with:" on the button "Change" you can open the menu to set the default PDF display program. There you will then most likely select the

"Adobe Acrobat" program if you have it installed.

🛃 Eigenscha	ften von BulkPDFBusiness_de.pdf X	
Allgemein Sic	berheit Details Vorgängerversionen	Wie möchten Sie .pdf-Dateien von jetzt an öffnen?
PDF		Diese App weiterhin verwenden
Dateityp: Öffnen mit:	Adobe Acrobat-Dokument (.pdf)	Adobe Acrobat DC
Ort:	Construction and Construction Control	Highlights in Windows 10
Größe:	9,05 MB (9.498.140 Bytes)	Microsoft Edge
Größe auf Datenträger:	9,05 MB (9.498.624 Bytes)	Öffnen Sie PDFs direkt in Ihrem Webbrowser.
Erstellt:	Samstag, 8. Januar 2022, 16:57:59	Weitere Optionen
Geändert:	Samstag, 8. Januar 2022, 16:58:00	Firefox
Letzter Zugriff:	Heute, 8. Januar 2022, 16:58:02	GIMP
Attribute:	<u>S</u> chreibgeschützt <u>E</u> rweitert	
Sicherheit:	Die Datei stammt von einem anderen Computer. Der Zugriff	
	wurde aus Sicherheitsgründen eventuell blockiert.	Dorfact DDE 10
		ОК
	OK Abbrechen U <u>b</u> ernehmen	

7.12 System.InvalidCastException: Could not convert data to boolean

This error occurs when you try to write a non-boolean value (yes or no) into a checkbox. Please read chapter 1.4.2 again for accepted values.

7.13 System.IndexOutOfRangeException: "Only X radio buttons in of the group. Selected index was Y."

This error occurs when a higher number is used for radio buttons than there are in total. For example, the error occurs if there are only 3 'Radio Buttons' in the 'Radio Button' group and you try to enter a 4 in the field. 1 to 3 would have been valid values, since a 'Radio Button' exists for this index. For 4, on the other hand, no 'Radio button' exists, which is why the error is displayed.

7.14 System.InvalidOperationException: The 'Microsoft.ACE.OLEDB.12.0' provider is not registered on the local machine.

This error occurs when the Microsoft Access Database Engine 2010 Redistributable 64 bit version is not installed on the computer. This can be done by Mi-

crosoft: http://www.microsoft.com/en-us/download/ details.aspx?id=13255

7.15 No data is recognized in the Excel file.

In this case, the necessary structure of the data is probably not maintained, for example, if there is an empty column or an empty row at the beginning of the table. A column is also considered empty if no column name is specified as text in the first row. Please take another look at the example

in section 1.3.2.

7.16 The date field is additionally specified with 00:00:00. How can this be removed?

To remove the time from the date, time/date formatting must be activated in the field and an appropriate formatting string must be inserted (see section 6.4). For example, the formatting string "dd.MM.yyyy" (case sensitive) results in "16.06.2022". The formatting string "MM-dd-yyyy" results in "16.06.2022".

"06-16-2022" and "dd\/MM\/yyyy" e.g. "16/06/2022". See section 3.3.8 for more information .

7.17 I have my data source files (Excel, CSV, etc.) or moved my PDF file. Are my BulkPDF files still working? Business configuration file?

Yes, but only after it contains a valid file path to the data source or PDF file. When opening the configuration file, BulkPDF Business will note that the file was not found and will allow you to select a new file path for the file. After that, the configuration will load normally again.

7.18 When calling the console application via a batch script, characters such as German umlauts are not transferred correctly

In this case it is necessary to define the "Code Page" in the batch script.

For German umlauts, for example, this would be possible with the command "chcp 1252" which should be at the beginning of the script. For more information on "Code Page" see https:// ss64.com/nt/chcp.html.

7.19 Combinations of multiple columns for one field

In BulkPDF Business it is not possible to use a function to merge data fields directly. However, you can use Excel functions in your XLSX file. For example, to merge data, you can use the Excel function CONCAT, which combines several cells into one

single cell. However, please note that the function name may vary depending on the system. After you have combined the desired data into a new column, it can be used in BulkPDF Business as a source

for a PDF form field.

7.20 "Checkbox" or "Radio button" should be according to data and preview in the XFA PDF, but in Acrobat Reader

This is due to the limited compatibility with XFA. However, it is possible to make the elements read-only for all

other PDF viewers. After this adjustment the elements are correctly selected, but can no longer be changed. Currently we do not know of any method to pre-select these XFA elements to a specific value if it is an XFA document.

7.21 How can I change the number format in my generated PDF correct it? It does not maintain the format set in the table. How can you, for example, enter numbers with two decimal places and a EUR symbol at the end?

Excel stores numbers as pure values, with additional details such as the euro symbol considered metadata. One way to specify your desired format in BulkPDF Business is to convert your number into a text field. This can be achieved with the Excel function =TEXT(A1;"#.##0,00 ") For more information about Excel function and format option, see https:// support.microsoft.com/en-gb/office/text-function-20d5ac4d-7b94-49fd-bb38-93d29371

8

References

- [1] BulkPDF website https://bulkpdf.de/
- [2] .NET Date Formatting https://

docs.microsoft.com/en-us/dotnet/standard/base-types/ custom-date-and-time-format-strings

[3] How To Create Fillable PDF Forms With LibreOffice Writer

https://www.linuxuprising.com/2019/02/how-to-create-fillable-pdf-forms-with. html

- [4] The original online Free PDF editor & form filler. https://www.pdfescape.com/
- [5] JavaScript Object Notation https://en.wikipedia.org/wiki/JSON

[6] Absolute and relative paths - Wikipedia

https://en.wikipedia.org/wiki/Path_(computing)#Absolute_and_ relative_paths

[7] ExcelDataReader

https://github.com/ExcelDataReader/ExcelDataReader